

Multitrait GWAS to connect disease variants and biological mechanisms

Hanna Julienne, Vincent Laville, Zachary Mccaw, Zihuai He, Vincent Guillemot, Carla Lasry, Andrey Ziyatdinov, Cyril Nerin, Amaury Vaysse, Pierre Lechat, et al.

▶ To cite this version:

Hanna Julienne, Vincent Laville, Zachary Mccaw, Zihuai He, Vincent Guillemot, et al.. Multi-trait GWAS to connect disease variants and biological mechanisms. PLoS Genetics, 2021, 17 (8), pp.e1009713. 10.1371/journal.pgen.1009713 . pasteur-03526210

HAL Id: pasteur-03526210 https://pasteur.hal.science/pasteur-03526210

Submitted on 14 Jan2022

HAL is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.



Distributed under a Creative Commons Attribution 4.0 International License



GOPEN ACCESS

Citation: Julienne H, Laville V, McCaw ZR, He Z, Guillemot V, Lasry C, et al. (2021) Multitrait GWAS to connect disease variants and biological mechanisms. PLoS Genet 17(8): e1009713. https://doi.org/10.1371/journal.pgen.1009713

Editor: Ron Do, Icahn School of Medicine at Mount Sinai, UNITED STATES

Received: January 20, 2021

Accepted: July 12, 2021

Published: August 30, 2021

Peer Review History: PLOS recognizes the benefits of transparency in the peer review process; therefore, we enable the publication of all of the content of peer review and author responses alongside final, published articles. The editorial history of this article is available here: https://doi.org/10.1371/journal.pgen.1009713

Copyright: © 2021 Julienne et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All GWAS summary statistics data used in this study are publicly available. Links to each dataset are provided in S1 Table. All other derived data are available in the manuscript or its supporting information files.

RESEARCH ARTICLE

Multitrait GWAS to connect disease variants and biological mechanisms

Hanna Julienne^{1*}, Vincent Laville¹, Zachary R. McCaw², Zihuai He³, Vincent Guillemot¹, Carla Lasry¹, Andrey Ziyatdinov⁴, Cyril Nerin¹, Amaury Vaysse¹, Pierre Lechat¹, Hervé Ménager¹, Wilfried Le Goff⁵, Marie-Pierre Dube^{6,7}, Peter Kraft^{2,4}, Iuliana Ionita-Laza⁸, Bjarni J. Vilhjálmsson^{9,10}, Hugues Aschard^{1,4*}

 Department of Computational Biology, Institut Pasteur, Paris, France, 2 Department of Biostatistics, Harvard TH Chan School of Public Health, Boston, Massachusetts, United States of America, 3 Department of Neurology and Neurological Sciences, Stanford University School of Medicine, Stanford, California, United States of America, 4 Department of Epidemiology, Harvard TH Chan School of Public Health, Boston, Massachusetts, United States of America, 5 Sorbonne Université, INSERM, Institute of Cardiometabolism and Nutrition (ICAN), UMR_S 1166, Paris, France, 6 Université de Montréal Beaulieu-Saucier Pharmacogenomics Centre, Montreal Heart Institute, Montreal, Canada, 7 Université de Montréal, Faculty of Medicine, Department of medicine, Université de Montréal, Montreal, Canada, 8 Department of Biostatistics, Columbia University, New York, New York, United States of America, 9 National Centre for Register-based Research, Department of Economics and Business Economics, Aarhus University, Aarhus, Denmark, 10 Bioinformatics Research Centre, Aarhus University, Aarhus, Denmark

* hanna.julienne@pasteur.fr (HJ); hugues.aschard@pasteur.fr (HA)

Abstract

Genome-wide association studies (GWASs) have uncovered a wealth of associations between common variants and human phenotypes. Here, we present an integrative analysis of GWAS summary statistics from 36 phenotypes to decipher multitrait genetic architecture and its link with biological mechanisms. Our framework incorporates multitrait association mapping along with an investigation of the breakdown of genetic associations into clusters of variants harboring similar multitrait association profiles. Focusing on two subsets of immunity and metabolism phenotypes, we then demonstrate how genetic variants within clusters can be mapped to biological pathways and disease mechanisms. Finally, for the metabolism set, we investigate the link between gene cluster assignment and the success of drug targets in randomized controlled trials.

Author summary

Genome-wide association studies (GWAS) established numerous associations between genetic variants and human traits. The anonymized summary of GWAS results is generally made publicly available to the scientific community and can be explored further. Amongst the many possible secondary analyses, one is to study the effect of a genetic variant on several traits (multi-trait GWAS) rather than a unique trait. We compared several tests to conduct multi-trait GWAS on simulated and real data. We detected 322 new associations that were not previously reported by standard univariate GWAS. We then detected clusters of genetic variants having a similar effect across several traits. Focusing **Funding:** This work has been conducted as part of the INCEPTION program (ANR-16-CONV-0005) (HA). It was also supported by NIH grant R03DE025665 to HA. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

on two subsets of immunity and metabolism traits, we demonstrate how genetic variants within clusters can be mapped to biological pathways and disease mechanisms. Finally, for the metabolism set, we investigate the link between gene cluster and success of drug targets in randomized controlled trials. We propose this method for improving the functional interpretation of GWAS results.

Introduction

Genome-wide association studies (GWASs) have identified thousands of significant genetic associations for multiple traits and diseases[1]. Publicly available summary statistics from these GWASs have proven to be invaluable in human genetic studies, enabling a range of secondary analyses without requiring individual-level genotype data and thus, averting major practical and ethical issues[2]. Among others, the estimation of phenotype heritability[3], the derivation of polygenic risk score^[4], and the assessment of causal relations between phenotypes^[5] are paragons of their critical utility. GWAS summary statistics have also been extremely useful to investigate pleiotropy and genetic correlation between human phenotypes. For example, recent works assessed whether significant loci for a given phenotype are also associated with other traits [6, 7] while others estimated genome-wide [8, 9] and regional [10] genetic correlations among phenotypes. The joint test of multiple traits can also be an efficient way to detect genetic variants missed by univariate screening [11-23], especially those with association patterns that deviate from the observed phenotypic correlation[24-26]. Nevertheless, while simulation studies and examples from real data applications in best case scenarios have confirmed the relevance of multitrait association tests, they have seldom been applied to large-scale datasets.

The application of multitrait association tests to a large heterogeneous set of traits requires overcoming several practical issues including careful preprocessing of individual GWAS summary statistics to avoid statistical artifacts, the estimation of multiple global parameters, and addressing widespread missing summary statistics. We addressed these issues in recent studies, developing the RAISS^[27] approach for imputation, and JASS preprocessing and multitrait analysis pipeline software [28]. Nevertheless, the relative performances of existing multitrait tests in real data have not been fully addressed. In brief, two types of methods have been developed, weighted sum of univariate statistics, assuming a specific multivariate genetic effect distribution[12, 14, 15], and an omnibus approach, allowing for one degree-of-freedom per statistic [11, 29], with some approaches using a combination of both [30]. An extensive and fair comparison of these methods is challenging as most face some of the aforementioned practical issues, no readily available implementation [28], and power in real data highly depends on the true (and unknown) multitrait genetic effect distribution. Finally, in addition to the potential ability to detect new variants, there is increasing interest in using GWAS multitrait association to decipher inter- and intra-phenotype genetic architecture[31, 32]. Again, real data applications are scarce, and questions remain regarding the approach to be used, the detectability of the multitrait genetic structure behind genome-wide genetic correlation, their potential matching to biological mechanisms, and their potential clinical utility.

Here, we build on previous works to conduct a large-scale multitrait analysis using GWAS summary statistics from 36 phenotypes categorized into five clinically relevant sets (*Immunity*, *Anthropometry*, *Metabolism*, *Cardiovascular* and *Brain*). We implemented five tests, an omnibus *K* degree freedom test (for *K* GWAS analyzed jointly) similar to [11, 29], and a weighted approach using four alternative weighting schemes, including some partly similar to previously

proposed ones[12, 14]. Comparing the relative performances of these models, we found substantial specificity of the signal identified by each approach, in terms of both association patterns and expressed tissue enrichment. We then used a Gaussian mixture model on the phenotypes with a variant association matrix to identify potential clusters of variants displaying similar genetic multitrait association profiles. In-depth functional analysis of the resulting clusters demonstrates a connection between those profiles and tissue specific expression. This breakdown of multitrait association signals highlighted how the overall genetic correlation between phenotypes can be decomposed into likely distinct genetic pathways. Finally, we used the phenotypes from the *Immunity* and *Metabolism* sets as case studies to demonstrate the matching between the identified profile and known biological pathways. Notably, mapping SNPs with unknown functions to pleiotropy profiles can indicate putative pathways. We conclude by investigating the potential clinical utility of the identified clusters for drug targeting.

Results

Multitrait genetic association test

The first step of our study consisted of ensuring the validity of the proposed statistical tests by studying potential bias, and assessing their statistical power under several simulation scenarios. Let us denote the vectors of single nucleotide polymorphism (SNP) Z-scores $\mathbf{z} = (z_1, \dots, z_K)$, where K is the number of phenotypes (i.e., the number of GWASs analyzed jointly). The first model we used, which we refer to as *sumZ*, assumes that genetic effects across the phenotype analyzed follow a direction specified by a vector of weights w to form a weighted sum of Zscores. Here we considered four weighting schemes: i) uniform weighting $(sumZ_1)$; ii) weighting according to the first principal component of the phenotypic correlation matrix $(sumZ_r)$; iii) weighting according to the first principal component of the genetic correlation matrix $(sumZ_g)$; and iv) weighting according to the independent component analysis of the Z-scores matrix ($sumZ_{ico}$). The second approach, which we refer to as *omnibus*, does not rely on a prior specification on the direction or magnitude of the SNP effect across traits. In brief, it compares, for one SNP, the vector of genetic effects \mathbf{z} with the expected multivariate normal distribution under the null. It is a standard omnibus test based on summary statistics that allows for one degree of freedom per outcome (here per phenotype). Both approaches (sumZ and omnibus) rely on a valid estimation of Σ_r , the variance-covariance matrix between z_1, \ldots, z_K , under the null hypothesis of no association, and share similarities with previous approaches (see S1 Text).

We first performed an in-depth validation of each approach, starting with a series of simulations under an ideal situation, when there were no missing data and the true Z-score covariance matrix, Σ_r is known (S1–S3 Figs). We further show using both simulated data (S4 and S5 Figs and S1 Text) and real data from the *UK Biobank* cohort that in the specific case of complete sample overlap between GWASs, the *omnibus* test is asymptotically similar to a multivariate analysis of variance (MANOVA) applied to individual level data (S6 Fig and S1 Text). The only major potential source of bias we identified is the misspecification of Σ_r which can lead to severe type I error inflation (S7 Fig). Misspecification can affect all variants, if Σ_r is estimated naively using the z-score data as proposed in previous studies (S8 Fig). Comparing various approaches, we found that Σ_r can be accurately estimated using *LDscore* regression[9] (S9 Fig), and the approach was therefore used to estimate Σ_r along the genome-wide genetic correlation (Σ_g) for the 36 phenotypes analyzed (S2 and S3 Tables). Nevertheless, misspecification can also be SNP-specific when sample size varies across the SNPs analyzed. Per-SNP sample size heterogeneity can induce different proportions of sample overlap and potentially invalidate Σ_r for those variants. We illustrate this potential bias by applying *omnibus* tests for the

joint analysis of the four GWASs from the GC consortium[33] (S10 Fig). To address this issue, we implemented additional tools to estimate sample size per SNP when missing and subsequently filtered the variants with small sample sizes (S11 andS12 Figs and S1 Text). Finally, out of 10 million variants reported for some GWAS, fewer than 1,000 had complete summary statistics for all 36 phenotypes analyzed. While methods exist to impute missing GWAS statistics, we found them to be inaccurate for multitrait analyses and we implemented the RAISS [27] approach we recently developed to ensure valid imputation for our context (S13 Fig). All preprocessing steps were recently incorporated into a publicly available toolset[28]. After applying our preprocessing pipeline to all 36 GWAS analyzed, there remained 6,978,319 SNPs with a missing data rate of 45% (59% before imputation).

Next, to illustrate the relative detection ability of each approach, we conducted a series of simulation studies across various scenarios. S14 Fig represents the null hypothesis rejection boundaries of each test using a simple example with two traits, when the genetic effect follows a single bivariate normal distribution and when it follows a mixture of two bivariate normal distributions (to reflect our working hypothesis). In this setting, the omnibus test displays the largest detection rate. Among the sumZ tests, $sumZ_{e}$ and $sumZ_{ica}$ show better performances than $sumZ_1$ and $sumZ_2$, especially when the structure of the genetic signal is heterogeneous. We then conducted more extensive simulations focusing on the *omnibus* and $SumZ_g$ tests. Fig 1 lays out the results of simulations using 10 traits generated under contrasting scenarios. Again, the omnibus test shows the best performances, especially when all traits have a high heritability and if the genetic signal is not structured along a specific direction. The statistical power of the *omnibus* test also tends to increase with the sample overlap especially when the environmental correlation was not aligned with the genetic correlation. Interestingly this good performance was also observed for genetically uncorrelated traits when they have a high heritability. In the case of a highly structured genetic effect sampled along a specific direction $SumZ_g$ and the *omnibus* test performed similarly. However, when only one of the 10 traits had a high heritability, the *omnibus* test underperformed compared to the $SumZ_{e}$, reflecting the cost of additional degrees of freedom in the omnibus test.



Fig 1. Statistical Power of the *omnibus* and *sumZ_g* tests under several simulation scenarios. Power of the *omnibus* and $sumZ_g$ tests with respect to sample overlap. Color of the line represents the test. Each panel correspond to a simulation scenario. Point shape indicates if the residual covariance was generated to be partially aligned with the genetic covariance or to be unconstrained (random). A 10 trait Z-score vector was generated as the sum of a genetic effect and a residual effect: $Z = Z_g + Z_{res}$. Z_{res} was sampled from a normal multivariate distribution with covariance matrix terms equal to $\sigma_z = \rho n_s / \sqrt{n_1 n_2}$ where n_1 is the sample size of the first study, n_2 is the sample size of the second study and ρ is the phenotypic covariance among the n_s overlapping samples. Z_g varied depending on the simulation scenario: (eff ~ Random) Z_g were sampled from a uniform distribution with boundary [-6; 6], (eff ~ corG) Z_g was sampled from a normal multivariate distribution simulating genetically uncorrelated traits with high heritability, (eff ~ het H) Z_g was samples from a normal multivariate distribution simulating genetically uncorrelated trait with only the first having a high heritability.

Comparison of multitrait association in real data

We analyzed the 36 GWASs of European ancestry (S1-S3 Tables) using the aforementioned multitrait approaches applied to seven phenotype sets: five medical-based sets (Immunity, Anthropometry, Metabolism, Cardiovascular and Psychiatric), a BMI related set including anthropometry traits and lipids (referred further as the *Composite* set), and finally all 36 phenotypes jointly (Fig 2). Note that we included bone mineral density traits in the Immunity set because an enrichment of BMD genome wide significant loci in immune pathways and immune cell regulatory regions has been previously reported [34, 35]. We derived the overlap of significant loci of the multitrait tests per phenotype set (S15-S21 Figs), and after merging all analyses (Fig 3A). We applied a Bonferroni correction to the joint tests and used a *p*-value threshold of 10^{-8} instead of the standard $5x10^{-8}$. Univariate phenotype associations were included in the comparison using the minimum of univariate p-value across all outcomes (noted P_{univ}). Across all phenotype sets, 391 associations were identified by the multitrait tests only, 392 were identified by univariate association tests only, and 1557 were significant for both univariate and multitrait tests (see Fig 3A). The largest number of new associations was detected by the omnibus test. The performances of the sumZ tests varied substantially depending on the phenotype set. For example, the weighting scheme based on phenotypic correlation $(SumZ_r)$, detects slightly more signals than other weights for the *Immunity* set (S19 Fig) but



Fig 2. Analysis overview. The diagram presents the overall analysis pipeline. A total of 36 GWAS were included covering several common diseases and quantitative traits. All GWAS summary statistics went through extensive preprocessing and quality control filtering, and missing single SNP statistics were imputed when possible. Multitrait approaches were then applied to all clean GWAS data and on each clinically based set (*All, Immunity, Metabolism, Brain, Cardiovascular, Anthropometry*, and *Composite*). After combining univariate and multivariate results, and merging SNPs within locus, a total of 6,767 associations were identified. After a comparison of results per approach, a clustering analysis was performed for variants within each set. Finally, we performed in-silico functional analysis of the clusters derived in the *Metabolism* set to assess their biological relevance.



Fig 3. Multitrait approach comparison. Panel (**A**) shows independent variants detected across the six approaches: univariate test (*univ*), omnibus test (*omni*), weight dsum of Z-score with uniform weight (*sumZ*₁), weight defined as the loading of the first principal component of the phenotypic correlation (*sumZ*_r), the genetic correlation (*sumZ*_g), or defined using the loadings of an independent component analysis (*sumZ*_{ica}). Each line corresponds to a test and each column to a set of significant variants. For each set, the test for which variants are significant are represented with a black dot on the test line. The barplot at the left represents the total number of significant independent signals detected by each approach. The stack bar at the top represents the cardinality of the sets. The next panels show the link between strengths of univariate association signal and the relative performance (i.e. larger power) of the four most tests: *univ, omni, sumZ*_g, and *sumZ*_{ica}, for each phenotype set: *anthropometry* (**B**), *cardiovascular* (**C**), *immunity* (**D**), *metabolism* (**E**), *brain* (**F**), *composite* (**G**), and *all phenotypes* (**H**). Within each phenotype set, we split the top associated SNPs per region based on the most significant test, and derived the median chi-squared for each test. The radar plots show the derived median per test and illustrate the strong heterogeneity in patterns identified. For example, out of the 1605 SNPs from the *anthropometry* set, 1235 had stronger signal with *univ* as compared with other tests. The median chi-squares in that group were 49.1, 1.1, 2.0, 1.0, and 0.7 for height, body mass index (BMI), hip circumference (Hip), waist circumference (WC), and waist to hip ratio (WHR). Comparatively, the 267 SNPs harboring a stronger signal with *omnibus*, had median of 6.8, 20.1, 15.9, 11.2, and 7.2 for the same phenotypes.

fewer associations in other phenotype sets (Fig 3A). While the *Omnibus* detected the largest number of new associations, the substantial share of signals found by other models suggests that applying several multivariate tests, especially the combination *omnibus*, *sumZ_{ica}*, *and sumZ_g*, could be an interesting solution to maximize detection. Finally, we checked the 392 associations identified by the multitrait test only in these data against previously reported associations from the GWAS catalog[1] for the same phenotypes. Altogether, we report a total of 322 new associations (S4–S10 Tables).

To further understand the relative performances of those three tests (*omnibus*, $sumZ_{ica}$, $sumZ_e$) along the univariate test, we explored which multitrait signal was associated with the largest increase in detection per test. To this end, we listed all loci found associated with at least one of the four approaches, and assigned each locus to a test based on the lowest *p*-value. We then derived the median squared z-score by phenotype across the loci assigned to each test. As shown in Fig 3B-3H, the median pattern varied substantially across tests and phenotype sets. Higher power for the univariate test was, as expected, observed for strong association signals for a single phenotype, and mostly reflected a very large sample size for that phenotype and/or a strong heritability (e.g. height in the anthropometry set, Fig 3B, or atrial fibrillation in the cardiovascular set, Fig 3C). A strong association signal for the *omnibus* test was linked to the inclusion of correlated phenotypes and sample overlap, resulting in a high residual covariance (Σ_r , S2 Table). For example, the median squared z-score was elevated for any stroke (AS), any ischemic stroke (AIS) and cardioembolic stroke (CES) in the Cardiovascular set. The patterns preferentially detected by the $sumZ_g$ test are harder to interpret. However, we noticed that sumZg displays a strong signal for SNPs associated with physiologically related traits (e.g., T2D and fasting glucose in the metabolism set, Fig 3E, or bone mineral density of neck and spine in the immunity set, Fig 3D).

To confirm the relevance of the associations detected by multivariate tests, we also conducted a tissue enrichment analysis to significant variants identified by the multitrait approaches and by the univariate analyses separately (**S11 and S12 Tables**). Overall, univariate variants and multitrait variants harbored a very similar functional enrichment landscape (**S22 Fig**). Most enriched tissues are already known to be involved in the phenotype in question, including liver, fat and pancreas for the *Metabolism* set, immune cell types and thymus for the *Immunity* set, and heart for the *Cardiovascular* set. Our enrichment study also confirmed less obvious observations, which have nevertheless been noted before: the involvement of immunity in brain-related traits (e.g. autisms and schizophrenia)[<u>36</u>, <u>37</u>] and the overrepresentation of brain tissues in the *Metabolism* set[<u>38</u>, <u>39</u>].

Distinct genetic association profiles correspond to distinct genetic correlations

Our comparison of approaches highlights that associated genetic variants display a broad range of multitrait association profiles. We investigated how these profiles can be broken down into groups of homogeneous multivariate genetic effects. This is directly related to the principle of genetic correlation, which quantifies the concordance of genetic effects across traits (e.g. [9]). The difference here, is that genetic correlation captures only the average over the whole genome, and as discussed in recent studies, more localized genetic structures likely exist for many pairs of traits[10]. To detect such a structure, we implemented a multivariate Gaussian mixture model (MGMM)[40] for the identification of clusters among SNPs found associated with at least one approach. We applied MGMM assuming between 2 and 10 clusters and used the BIC and silhouette criteria to determine the most relevant number of clusters. We further bootstrapped the computation of the clustering criteria to ensure the robustness of



Fig 4. Multitrait genetic association clusters for the *Metabolism* set. The panels summarize the clustering of the 392 independent SNPs selected from the *Metabolism* set analysis. The set includes 10 phenotypes: triglyceride (TG), total cholesterol (TC), type 2 diabetes (T2D), low-density lipoprotein cholesterol (LDL-C), high-density lipoprotein cholesterol (HDL-C), glycated hemoglobin (HbA1c), Homeostasis model assessment of β -cell function (HOMA-B), homeostasis model assessment of insulin resistance (HOMA-IR), fasting insulin, and fasting glucose. The alluvial plot in panel A) represents the decomposition of univariate genetic association and its rewiring to the six inferred clusters. The flow widths represent the proportion of phenotype's variance explained by the subset of SNPs assigned to each specific cluster, relative to the total genetic variance explained by all 392 SNPs. For example, SNPs from cluster 6 capture approximatively 41.7% and 54.6% of that genetic variance for TC and LDL, respectively. For clarity, flows explaining less than 0.1% of the variance are not represented. Panel B) shows the heatmap of normalized beta coefficients per phenotype within each cluster. Each column is a SNP, with blue and red colors indicating negative and positive beta, respectively. Coded alleles have been defined according to the per cluster first principal component. The boxplots in panel C) shows the distribution per cluster of SNP's entropy, an indicator of the fitness of the SNP-cluster assignment. SNPs perfectly assigned are expected to have entropy close to zero.

https://doi.org/10.1371/journal.pgen.1009713.g004

the selection (**S1 Text**). The best suited number of clusters is 6, 8, 8, 9, 3, 2 and 5 for the *Metabolism*, *Immunity*, *Cardiovascular*, *Anthropometry*, *Psychiatric*, *Composite*, and *All* sets, respectively (**S23 Fig**). As illustrated for the *Metabolism* set in **S24 Fig**, adding significant SNPs from the multitrait tests to those identified by the univariate tests enabled us to detect more clusters.

We assessed the uncertainty in cluster assignation by deriving the entropy for each variant (see **S1 Text**). We observed some heterogeneity in the distribution of entropy values across phenotype sets and clusters (**Fig 4C** and **S16** and **S17 Tables**). The difficulty to attribute a cluster for certain variants might be due to the lack of representative cluster (*i.e.* sub-structure not captured in our analysis) or to shared functionalities between clusters that are not modelled in the GMM framework. Because differentiating those two possibilities using the available data would be very challenging, we decided to remove outlier variants (N = 28 across all sets) with entropy above 0.75 from further analyses.

The resulting clustering is presented in Fig 4 for the *Metabolism* set and in $\underline{S25}$ -S31 Figs for the other sets. Each figure presents a heatmap of Z-scores along with an alluvial plot displaying



Fig 5. Heterogeneity of genetic correlation across clusters for the *Metabolism* **and** *Immunity* **sets.** We derived the genome-wide genetic correlation between phenotypes using *LDscore* regression and using Pearson correlation from all SNP Z-scores (top panels), and for SNPs within the identified clusters. Results for the *Metabolism* set are presented in panel (**A**) using only the four key traits, LDL, HDL, Triglyceride (TG) and type 2 diabetes (T2D). Results for the *Immunity* set are presented in the panel (**B**). For clarity only significant correlation are represented. The boldness of the line is proportional to the strength of the genetic correlation. Positive correlations are represented in blue and negative correlations in red. The values of the genetic correlation are indicated by the number next to the trait. Solid lines represent significant correlation (after Bonferroni correction) whereas dashed lines represent correlation significant only before Bonferroni correction. Note that because the clusters are inferred from the multivariate associations, the absolute value of the significance of the correlations is of limited interest. Nevertheless, it provides a useful descriptive statistic to identify the key structures within each cluster.

https://doi.org/10.1371/journal.pgen.1009713.g005

both the shared explained variance between phenotypes and the proportion of explained variance by clusters for each phenotype. The complete list of SNPs for the *Metabolism* set per cluster is presented in **S16 Table**. The multivariate effects vary substantially from one cluster to another. For instance, in *Metabolism* clusters, SNPs from the cluster 3 display increased HDL-C and decreased triglycerides, while SNPs from cluster 5 are more specific to triglycerides.

The alluvial figures and heatmaps provide an overview of the magnitude of genetic effects from one cluster to another. To further characterize the concordance or discordance of genetic contributions across phenotypes, we computed the pairwise SNP-based genetic correlations for each cluster (see <u>S1 Text</u>). Fig 5 presents those estimates for a subset of phenotypes within the *Metabolism* and *Immunity* phenotype sets. In the *Immunity* set, the correlations between

rheumatoid arthritis (RA), ulcerative colitis (UC) and Crohn's disease (CD) provide a striking illustration of how the genome-wide genetic correlation can be composed of smaller structures. The genome-wide genetic correlations between UC and CD is strong (0.41), but near 0 and not significant for RA (see S3 Table). In Fig 5B, we noticed a fairly large negative correlation in clusters 2 and 3 between RA both CD and UC whereas, the cluster 5 captures a group of variants displaying a strong positive correlation across the three traits. Similar negligible genome-wide correlations along with opposite genetic correlations across clusters were observed in the *Metabolism* set. For example, variants from cluster 1 display a strong negative correlation. Fig 5 also highlights that significant genome-wide genetic correlations across highly related phenotypes such as UC-CD and LDL-TG are not distributed evenly across variants.

Biological meaning of genetic clusters

We conducted series of *in silico* functional analyses with the objective of mapping clusters to candidate biological functions. For each phenotype set, we evaluated two types of enrichment: tissue-specific chromatin mark enrichment per cluster (S13 Table) and a pathway enrichment framework (S14 and S15 Tables) which integrates multiple databases such as Gene Ontology (GO) and KEGG. Here, we focused on the *Immunity* and *Metabolism* sets as a case study. Given a phenotype set, while using the same set of traits for all clusters, we observed large differences in pathways, tissues and cell type enrichments between clusters.

For the Immunity set, clusters 1 and 4 predominantly capture genetic effects of bone-mineral density; clusters 2, 3 and 5 affect inflammatory bowel disorder (IBD); and clusters 6, 7 and 8 capture variants with pleiotropic effects on rheumatoid arthritis and IBD (\$27 Fig). Both enrichment analyses pointed toward an overrepresentation of the immune system with all clusters-even those affecting primarily bone-mineral density-being enriched for at least one immunologic pathway or one immunological tissue. We highlight the top enriched tissues and top pathways in Table 1. Concerning pathway enrichment, immune related pathways regulating the shape of the immune response, such as cytokines and the JAK-STAT signaling pathway were recurrent. Interestingly, variants from those clusters map to a distinct set of cytokines and cluster of differentiation genes (e.g., IL4, IL13, IL33 for cluster 1 and IL3, IL5, IL10, IL19, IL20, IL21, IL27 for cluster 5), which suggests that they may impact different components of the immune system. Concerning tissue-specific active chromatin mark enrichment, clusters 2 and 3 contained multiple SNPs enriched primarily in transcriptionally active regions of "primary natural killer cells from peripheral blood" whereas cluster 7 and 8 are enriched for "primary T helper cells." We also observed enrichment in the tissue where immune damage occurred for cluster 5 (colonic mucosa), which highlights the complex interaction between the immune system and the inflamed tissue.

The *Metabolism* set includes several molecular phenotypes, which we expect to be closer to biological mechanisms than some of the macrophenotypes from other sets. Overall, cluster 1 is mostly associated with an increase in fasting glucose and impaired β -cell function; cluster 2 is highly pleiotropic and notably increased the risk of T2D; and clusters 3 to 6 are mostly associated with lipids (Fig 4). Accounting for the direction of effects, we also noted that the genetic associations in cluster 5 match the known phenotypic correlation with the inverse relationship between circulating levels of HDL-C with those of LDL-C and, more especially, TGs observed in epidemiological studies[41]. At the tissue level, we observed modest enrichment for adipocytes in clusters 1 and 2 (FDR *p*-value 0.028 and 0.01 respectively, S13 Table) and cluster 3 SNPs are up-regulated in the liver (FDR *p*-value 0.005).

		1			, , ,		
Cl.	#SNP ^a	#gene ^b	Top GTEx Tissue	(q-val)	Top Immunologic pathways	(q-val)	Immunity related genes
1	32	55	-	-	http://www.gsea-msigdb.org/gsea/msigdb/geneset_page.jsp? geneSetName=GOMF_CYTOKINE_ACTIVITY&keywords= GOMF_CYTOKINE_ACTIVITY	(1.9 x10 ⁻³)	IL4, IL13,IL33, STAT6, TNFSF11, TSLP, FAM3C, TNFRSF11B
2	40	55	Primary Natural Killer cells from peripheral blood	(6.2x10 ⁻⁵)	KEGG_JAK_STAT_SIGNALING_PATHWAY	(1.9 x10 ⁻⁹)	IL10, IL12B, IL3, IL4, IL5, IL13, IL19, IL3, IL12RB2, IL23R, CSF2
3	83	190	Primary Natural Killer cells from peripheral blood	(2.8x10 ⁻⁴)	KEGG_JAK_STAT_SIGNALING_PATHWAY	(6.2 x10 ⁻⁷)	IL3, IL26, IFNG,IL12RB2, IL17REL, IL23R, IFNGR2, CD244, CD274, STAT5A, STAT3, LIF, OSMR, CSF2, CCL13, CCL1, TNFSF15, TNFSF8, JAK2
4	39	96	Bone Marrow Derived Cultured Mesenchymal Stem Cells	(3.5 x10 ⁻²)	KEGG_JAK_STAT_SIGNALING_PATHWAY	(0.020)	IL2, IL21, IL1R1, IL1RL2, CSF3, STAT3, SPRY1, TSLP
5	170	430	Colonic Mucosa	(7.5 x10 ⁻⁵)	https://www.gsea-msigdb.org/gsea/msigdb/geneset_page.jsp? geneSetName=IMMUNE_SYSTEM_PROCESS&keywords= IMMUNE_SYSTEM_PROCESS	(1.8 x10 ⁻⁸)	IL3, IL5, IL10, IL19, IL20, IL21, IL27, IL12RB2, IL18R1, IL1R2, IL1RL1, IL23R, CD19, CCL2, CCL7, CCL11, NOD2, TNFRSF9, JAK2
6	90	198	-	-	https://www.gsea-msigdb.org/gsea/msigdb/geneset_page.jsp? geneSetName=IMMUNE_RESPONSE&keywords= IMMUNE_RESPONSE	(1.3 x10 ⁻⁷)	ILF3, IL12RB2, IL18RAP, IL23R, CD28,CD40, C5, STAT4, STAT1, TYK2
7	20	18	Primary T helper naive cells from peripheral blood	(7.5 x10 ⁻⁵)	-	-	
8	121	59	Primary T helper memory cells from peripheral blood 2	(2.5 x10 ⁻⁴)	-		IL6R, TNFAIP3

Table 1. 1	Fop tissue associations and	Immune related Genes b	y Clusters for the Immunity	y set
------------	-----------------------------	------------------------	-----------------------------	-------

Cluters (Cl.) not mapping to neither tissues nor pathways are indicated by a "-" sign. All reported p-value are FDR corrected.

^a Count includes only the most associated SNP per region.

^b Count of genes mapped to SNPs.

https://doi.org/10.1371/journal.pgen.1009713.t001

As shown in **S14 Table**, each cluster was significantly enriched for a large number of GO terms. We report some specific and illustrative examples: cluster 1 is enriched for the carbohydrate homeostasis set (q-value = 2.5×10^{-3}), cluster 3 is enriched for the reverse cholesterol transport set (*q*-value = 2.8×10^{-13}), cluster 4 is enriched for the plasma lipoprotein clearance set (q-value = 1.7×10^{-5}), cluster 5 is enriched for the protein lipid complex assembly set (qvalue = 1.08×10^{-9}) and cluster 6 is enriched for the low density lipoprotein particle remodeling set (q-value = 1.07×10^{-2}). Cluster 4 also exhibits active chromatin tissue enrichment in immune T cells (q-value = 2.3×10^{-3}), highlighting the link between cholesterol and immunity. Indeed, cholesterol and modified forms of cholesterol, such as oxidized cholesterol and cholesterol crystals, promote inflammatory and immune responses through multiple pathways including the activation of the Toll-like receptor (TLR) signaling, the NLRP3 inflammasome and myelopoiesis[42, 43]. While the promotion of inflammation and immunity is carried by LDL particles, HDL particles were proposed to counteract this effect in part through reverse cholesterol transport[44]. However, cluster 3, which is enriched for reverse cholesterol transport did not exhibit such tissue enrichment in immune T cells, indicating that the link between HDL and immunity may be more complex, as recently pointed out by Madsen et al[45].

Metabolism pathways and diseases

To provide a perspective on the specificity of genetic variants across clusters and their potential contribution to human diseases, we investigated the lipids from the *Metabolism* set. We first projected each cluster gene onto KEGG pathways. Here, we used only maps corresponding to enriched GO gene sets identified or to tissue identified in the enrichment analysis at the previous stages (**S14 and S15 Tables**): fat digestion and absorption, cholesterol metabolism, and

PPAR signaling pathways. We constructed a synthesis of these observations on the metabolic map presented in Fig 6A and 6B. Genes associated with clusters (S16 Table) had functions in agreement with their effects on blood lipid levels: cluster 3 is enriched in genes involved in HDL-C biogenesis and metabolism (*LCAT*, *ABCA1*, *SR-B1*, *CETP*, *PLTP*, *LIPG*, *APOAx and APOCx*), clusters 4 and 6 with genes related to LDL-C clearance (*SORT1*, *PCSK9*, *LDLR*, *LDLRAP1*, *APOB and APOE*), and cluster 5 to genes related to triglycerides and chylomicron transport (*LPL*, *APOAx and APOCx*).

We then assessed the effect of variants from each cluster with three diseases known to be associated with serum lipids: coronary artery diseases (CAD), stroke, and obesity (defined as a BMI > 30) (S19 Table). Within each cluster, we aligned the SNP alleles with the main trend of the corresponding cluster so that all coded alleles fit the multitrait pattern defined in Fig 6C (see S1 Text). For example, all SNPs from cluster 5 were recoded to be associated with an increase in TGs, TC and LDL-C and a decrease in HDL-C. We plotted in Figs 6D and S31 the genetic effect of each SNP on the three diseases (using the effect on BMI as a proxy for obesity) after the aforementioned alignment, and we performed a sign test to assess the significance of the observed trend (S19 Table). SNPs from several clusters displayed a significant increase in the risk of CAD: cluster 2 ($P = 6.6 \times 10^{-5}$), cluster 4 ($P = 2.9 \times 10^{-2}$), cluster 5 ($P = 3.9 \times 10^{-3}$) and cluster 6 ($P = 2.8 \times 10^{-4}$). SNPs from cluster 2 also displayed a nominally significant increase in the risk of stroke (*P*-value = 1.6×10^{-2}). Finally, a large fraction of SNPs from cluster 3 had a negative effect on BMI ($P = 6.4 \times 10^{-4}$). Interestingly, several SNPs from this cluster show association with CAD, but with heterogeneous effects-some associated with an increased risk and other associated with a decreased risk-so the absence of a global trend. The associations of clusters 4 and 6 with CAD add to the evidence of a causal effect of LDL-C on CAD[46], which has been established by prospective epidemiological studies[47], Mendelian randomization [48] and randomized clinical trials evaluating the effect of LDL-C reducing therapies[49]. Moreover, the association of cluster 5 with CAD risks corroborates a potential causal role of TGs[5] and remnant cholesterol[50, 51] in CAD. The role of TGs in CAD has also been confirmed by epidemiological studies[52], genome-wide association studies[5], Mendelian randomization studies^[53] and randomized controlled trials aiming the lower of TGs^[54]. Cluster 3, which is associated with increases in HDL-C, does not have a protective effect on CAD, which is again in agreement with Mendelian randomization analyses reporting no link between HDL and CAD[48, 55]. Finally, the association of cluster 2 with CAD and strokes further supports the potential causal effect of type 2 diabetes on CAD and stroke[56].

As a final exploratory analysis, we reported the cluster and multitrait genetic effect of genes targeted to mitigate hyperlipemia to prevent CAD (Table 2). Drug targets corresponding to cluster 3 (ABCA1, CETP, NR1H3) did not lead to successful clinical trials, whereas targets (PCSK9, NPC1L1, APOC3, HMGCR) in clusters 4, 5 and 6 were mostly successful or promising. The example of the CETP gene, that is classified in cluster 3, a cluster not associated with CAD, is of particular interest. CETP has been the target of failed clinical trials that attempted to prevent CAD by inhibiting CETP and consequently increasing circulating HDL-C[57–59]. Cholesteryl ester transfer protein (CETP) promotes the heteroexchange of cholesteryl esters and TGs between HDL-C and APOB-containing lipoproteins connecting HDL-C and TG metabolism[57]. Pharmacological inhibition of CETP was motivated by GWAS[60] and prospective cohorts[61] that indicated that CETP variants were associated with higher circulating HDL-C levels and lower LDL-C, TGs and CVD risk. However, although all CETP inhibitors achieved an effective increase in HDL-C, only anacetrapib led to a significantly lower incidence of major coronary events[62] in patients who were receiving statin therapy, an effect that might account for the reduction in ApoB (non-HDL-C) rather than the elevation of HDL, as suggested by Mendelian randomization analyses[63]. In addition to these well-known



Fig 6. Mapping clusters to pathways. We projected cluster's genes from the *Metabolism* phenotype set onto KEGG pathways and reconstructed a synthetic metabolic map. Panel **A**) presents the results for the lipoprotein component and panel **B**) for the lipid component. Gene names are highlighted by the colors of their associated clusters. When a gene is associated to several SNPs belonging to different clusters it is represented with several colors. To improve interpretation, we also present in panel **C**) a proxy for the relative contribution of each phenotype per cluster, defined as the loadings of the first principal component derived from the matrix of Z-score for the subset of SNPs in that cluster. Finally, panel **D**) shows the distribution of standardized beta for association between SNPs from each cluster and three diseases: any stroke (AS), coronary artery disease (CAD), and obesity (using body mass index as a proxy).

Target ^a	Drug (phase)	rsID ^b	Clu.	SNP-phenotype association ^c							Comment
				HDL	LDL	TC	TG	CAD	AS	BMI	
ABCA1	Probucol (4)	rs11789603	3	7.70	1.6	4.66	2.07	1.42	0.25	-1.50	This LDL-c lowering drug was approved but subsequently discontinued because of its lowering effect on HDL-c
CETP	Cetrapid (4)	rs12448528	3	27.79	-4.61	4.96	-4.60	0.25	-0.28	1.21	Three clinical trials were halted because they showed adverse effect and/or no therapeutic efficacy, except in the case of anacetrapid use for preventing new acute coronary events in high-risk individuals.
NR1H3	HDCA (1)	rs12575609	3	9.11	-0.19	1.89	-3.26	0.76	0.06	-3.9	The RCT results were not produced due to AtheroNova Inc. bankruptcy.
PCSK9	alirocumab, Evolocumab (4)	rs7523242	4	-1.16	10.49	9.28	1.92	3.25	1.03	-0.29	Approved second line treatment for high cholesterol individuals whose cholesterol is not controlled by Statin alone.
NPC1L1	Ezetimibe (4)	rs217386	4	-0.80	6.60	5.96	2.44	2.19	0.86	-1.47	Currently used to lower the absorption of cholesterol and is often used in association with statin.
APOC3, APOA1	Volanesorsen (3)	rs1815787	5	-2.09	5.45	9.41	16.60	0.39	0.26	0.614	A triglyceride-reducing drug currently in phase 3 RCT.
HMGCR	Statins (4)	rs59014134	6	0.79	15.79	16.06	1.34	2.01	-0.36	-4.59	The most common cholesterol lowering drugs.
АРОВ	Mipormersen (4)	rs1041968	6	-6.96	22.94	20.92	9.38	2.45	-1.20	-1.75	Can be used for risk management in familial hypercholesterolemia but can cause fatty liver disease.

Table 2. Drug target genes and associated SNPs in the metabolism set

^aNote that for probucol, the molecule inhibit ABCA1, but is not specific to ABCA1.

^bPrimary associated SNP and corresponding cluster. But note that for several loci, there is a few other SNPs from other cluster.

^cDefine as the association Z-score for the most associated variant in the gene.

https://doi.org/10.1371/journal.pgen.1009713.t002

drugs, we provide a systematic listing of potential drug targets by cluster (<u>S20 Table</u>) based on the druggable genome database[64].

Altogether, these results suggest that drug development might be more effective by accounting for the gene context, *i.e.*, by selecting candidate genes not from their individual features but based on the disease association trend of genes displaying similar multitrait association profiles. Under this working hypothesis, the proposed inference of genetic functional groups can provide a means to identify those genes and therefore to select potential candidates.

Discussion

In this study, we conducted multitrait analyses of GWAS summary statistics from 36 human phenotypes combining association tests and clustering to detect the shared and specific genetic substructures underlying those phenotypes and to explore the links between those substructures and biological pathways and diseases. The question of substructures underlying genomewide genetic correlation has been partially explored in other recent studies [8, 10]. Our work is in agreement with these studies, confirming the presence of regional genetic correlation differences and offering a data-driven approach for identifying primary substructures across millions of possibilities. Nevertheless, an understanding of whether those genetic functional groups are only statistical constructs or correspond to meaningful biological mechanisms is critical. In the latter, it means that a data-driven approach, such as the one proposed in the present study, can be used to dissect the genetic contribution of complex human phenotypes. Here, we used two complementary functional enrichment analyses to map these multitrait association profiles to pathways, and we report a detailed view of these profiles for the Immunity and Metabolism phenotype sets. In these analyses, the most enriched tissues varied substantially across clusters within each phenotype set, highlighting the potential of such an approach for characterizing distinct genetic mechanisms.

The variability in pleiotropy profiles across SNPs identified in GWASs has been previously discussed. For example, earlier reports[33] on inflammatory diseases have highlighted such patterns, or proposed grouping of SNPs based on the direction of association[65]. However, those studies used only a handful of SNPs identified at the time of publication. Our analysis based on formal clustering and functional enrichment analyses and using the results of GWASs performed in much larger sample size offers a new and much more detailed qualitative perspective on these profiles. More recent publications have also discussed approaches focusing on the characterization of SNPs displaying pleiotropic effects [66], the inference of shared and distinct genetic pathways between related phenotypes^[67], and the identification of genetic components linked to disease subtypes[31, 68]. Our approach shares objectives with some of these methods but also has unique features and advantages. Studies based on component decomposition techniques alike principal component analysis[31, 67] yields endotypes of interest from a biological standpoint, but do not provide the SNP-level genetic decomposition that we are addressing. Approaches that rely on individuals' genotypes are limited by the ethical and practical cumbersome aspects tied to this type of data[68], sometimes without increasing statistical power. For example, as demonstrated in this study, the MANOVA derived from individual-level data is equivalent to the proposed omnibus test derived using the summary statistics only.

Past studies have shown that sufficiently curated genetic information can enhance the chance of success of clinical trials[69, 70]. We further argue that fine analysis of pleiotropic effects, as performed in the present study, is a very promising path forward to help identify drug targets with a minimal risk of serious side effects. In particular, the picture of the links between coronary artery disease risk and lipid pathways inferred from our analysis is coherent with the state-of-the-art, while providing critical new evidence. While the association of LDL-C and TGs with CAD is largely documented [46, 71], the relation linking HDL-C with CAD is more complex, as both low and high HDL-C levels have been associated with a risk of cardiovascular disease and mortality [72, 73]. Recent studies pointed out that the functionality of HDL rather than the static measure of its circulating cholesterol level accounts for the relationship between HLD-C and CVD and mortality [73, 74], with a potential role of HDL in remnant cholesterol transport. Overall, evidence for the presence or absence of a causal effect between lipid cholesterol measures and CAD as reported by Mendelian randomization analyses should be considered with caution, as lipid traits result from a complex interconnexion of multiple biological pathways. Our analysis suggests that the genetic contribution to the established negative correlation between HDL-C and CAD might be driven only by a subset of genes within a few specific genetic pathways. Under this hypothesis, drugs targeting mechanisms outside these pathways would be ineffective in decreasing CAD risk. Note that such retrospective analysis can be only suggestive of the potential deleterious side effects of drug targets. Nevertheless, the identification of those candidates might limit cost of further analysis using both *in silico* analyses in independent data, and *ex vivo* study to examine the role of the identified genetic variants on additional intermediate molecular phenotypes (gene expression, protein, etc).

A number of further analyses can be conducted based on the results we obtained. First, we focused on a limited number of phenotype sets. Extending analyses to other sets of phenotypes might help refine potential genetic functional groups and better characterize their link to biological mechanisms. To our knowledge, there are no trivial solutions to solve the intrinsic combinatorial issue (i.e., one can build over $6x10^{10}$ sets of phenotypes from 36 GWASs). Additionally, note that we worked with a data freeze dated from December 2018. Hence, at the date of the publication of this paper, newer summary statistics are available for a few traits. We accounted for these new publications when counting newly identified variants by filtering

associations reported in the latest version of the GWAS catalog. Another critical component of our analysis is the methodological choices for clustering. Here, we considered a Gaussian mixture model, mainly to enable missing values, and used BIC and silhouette to decide the optimal number of clusters. Other methods and alternative criteria might result in slightly different clusters. Moreover, we assume that genetic variants belong to distinct clusters, but it is likely that some variants belong to multiple biological pathways. Note that GMM provides posterior probability of cluster assignment and has the potential to explore overlapping clusters, but better approaches might potentially exist to address that specific question. Additionally, our implementation does not automatically address the problem of allele coding (i.e., the choice of the coded allele) inducing, in some cases, symmetric clusters that we had to merge *a posteriori*. Again, alternative approaches might offer the possibility of solving this issue.

To summarize, we ensured the theoretical reliability of a panel of multitrait tests and demonstrated their capacity to detect new associations on diverse sets of traits. Considering independent significant associations, we stratified SNPs in multitrait profiles corresponding to biological pathways. We believe this stratification to be relevant for multiple applications ranging from functional annotation to drug targeting.

Online methods

Multivariate association test

Consider a vector **z** of *K Z*-scores statistics for a single nucleotide polymorphism (SNP) obtained from standard univariate genome-wide association screenings of *K* phenotypes. Under the null hypothesis, $\mathbf{z} = (z_1, ..., z_{K_2})$ follows a normal distribution $N(0, \Sigma_r)$, where Σ_r is the residual phenotypic covariance matrix (S1 Text), while under the alternative, **z** is expected to display additional covariance due to shared genetics (defined by a genetic correlation matrix Σ_g). We first considered an *Omnibus* test of the vector of Z-scores, which can be performed using the multivariate Wald statistics:

$$T_{omni} = \mathbf{z}^t \Sigma_{\mathbf{r}}^{-1} \mathbf{z}$$

where T_{omni} follows a chi-square with *K* degree of freedom (df) under the null hypothesis of no phenotype-genotype association. We also considered a classic weight-based test defined as:

$$T_{sumZ} = \frac{\left(\mathbf{w}^t \mathbf{z}\right)^2}{\mathbf{w}^t \boldsymbol{\Sigma}_{\mathbf{r}} \mathbf{w}}$$

where **w** is a vector of *K* weights applied to the *Z*-score. Under the null, T_{sumZ} follows a chisquared distribution with 1 degree of freedom. Note that this approach shares similarities with both standard fixed effect meta-analysis[14] and with dimensionality reduction methods (e.g. principal component analysis[25]). One can also note that the *Omnibus* statistics can be expressed as a combination of the *sumZ* statistics over all eigenvectors of Σ_r (S1 Text). We note **v**_i the ith eigen vector of Σ_r :

$$T_{omni} = \sum_{i=1}^{K} T_{sumZ} | \mathbf{w} = \mathbf{v}_i$$

We considered four weighting schemes for the *sumZ* tests: (i) in the SumZ₁, **w** is equal to the unit vector so all traits have the same weight; (ii) in the SumZ_r, **w** is equal to the first eigen vector of Σ_r so its direction represents phenotypic correlation between traits, (iii) in the SumZ_g, **w** is equal to the first eigen vector to Σ_g so its direction represents genetic correlation between traits, (iii) in the SumZ_g, **w** is equal to the first eigen vector to Σ_g so its direction represents genetic correlation between traits, (iii) in the SumZ_{ica} **w** is computed by applying an Independent component

analysis (ICA) to the complete matrix of Z-score. To compute the weight vector **w** of the Sum- Z_{ica} , for a given phenotype set, the genome wide Z-score matrix was extracted and an independent component analysis was performed with the scikit-learn python package. The component yielding the most novel association was selected as loadings. We verified that this selection procedure did not lead to an inflation under the null hypothesis by simulation (see <u>S2 Fig</u>).

Performing the omnibus test requires inverting the *Z*-score covariance matrix Σ_r . When this matrix does not have a full rank, we use a pseudo inverse of the matrix based on the singular value decomposition (S1 Text). Briefly, as Σ_r is a variance-covariance matrix, it can be written **PDP^t** where **D** = $diag((\lambda_k)_{k=1...K})$, $(\lambda_k)_{k=1...K})$ are the eigenvalues of Σ_r and **P** is the orthogonal matrix whose columns correspond to the eigenvectors of Σ_r . If it is not invertible, only *K*' eigenvalues are different from 0 (where *K*' denotes the rank of Σ_r) and an inverse Σ_r^{-1} of the matrix can be computed as $\Sigma_r^{-1} = \mathbf{P}_{K'} \mathbf{D}_{K'}^{-1} \mathbf{P}_{K'}^t$, where $\mathbf{D}_{K'}^{-1} = diag((1/\lambda_k)_{k=1...K'})$ and $\mathbf{P}_{K'}$ denotes the $K \times K'$ matrix whose columns are the *K*' eigenvectors corresponding to the eigenvalues different from 0. Note that the Omnibus statistics computed with Σ_r^{-1} follows a χ^2 distribution with *K*' degree of freedom.

Robust estimation of Z-score covariance

The validity of the proposed multivariate tests mostly relies on the accurate estimation of Σ_r . In practice, the covariance between Z-scores from null SNPs from two GWAS will deviate from 0 when there is both sample overlap and correlation among the traits analyzed. When combining results from two independent studies, or when the trait analyzed has negligible correlation, Σ_r will be a diagonal matrix, so that the *Omnibus* test can be performed by summing chi-squared statistics for each SNP to form a K degree of freedom chi-square, and the sumZ test becomes a standard weighted meta-analysis of fixed effect. Yet, in the large-scale GWAS era, this situation is unlikely as most of the large GWAS are conducted in the consortium setting, where samples likely overlap across multiple GWAS. It follows that Σ_r can contain nonzero off-diagonal terms. Under the complete null model, the expected Z-score covariance for null SNPs between two traits equals $\sigma_z = \rho n_s / \sqrt{n_1 n_2}$ where n_1 is the sample size of the first study, n_2 is the sample size of the second study and ρ is the phenotypic covariance among the n_s overlapping samples (see S1 Text and e.g. [3, 9]). In some specific cases, one can obtain these parameters directly from the data (e.g. when analyzing multivariate omics data). Conversely, obtaining all four parameters (ρ , n_s , n_1 , n_2) from consortium GWAS based on dozen or even hundreds of cohorts can be a practically daunting and risky task. Moreover, accurate phenotypic covariance estimation would be particularly challenging when study-specific and trait-specific covariates adjustment has been performed. Recent studies proposed to estimate $\Sigma_{\rm r}$ using available SNPs from the GWAS in question using all available single SNPs Z-score [75] or using a random subset of pruned variants[14], though some discussed removing GWAS hits[15], focusing on a subset of SNPs in regions less likely to contain causal variants [76], or using tetrachoric estimator [16]. The validity of these estimators mostly relies on the assumption that the vast majority of the SNP effects in the genome are distributed under the null hypothesis. While this is likely to be true in some cases, associated variants can potentially lead to either upward or downward pairwise covariance between Z-scores. Instead, we leverage recent work by Bulik-Sullivan et al[3, 9] that allows for estimation of this covariance (and the diagonal variance terms) under a polygenic model and assuming multivariate normality of effect sizes across traits (see S1 Text). The estimation of Σ_r was performed on Z-scores before the imputation step described in the next section. For a few traits the estimated variance is markedly inferior to 1. As indicated in the LDSC regression method, this phenomenon happens when the original GWAS was corrected with a genomic control factor.

Simulation studies on test statistical power

To assess the statistical power of the Omnibus test and the SumZ tests, we designed the following simulation scenarios. A 10 trait z-score vector was generated for each of the 5000 causal SNPs as the sum of a genetic effect and a residual effect: $Z = Z_g + Z_{res}$. In the following, random covariance matrices were generated using the randcorr R package[77]. Z_{res} was sampled from a normal multivariate distribution with covariance matrix (Σ_r) terms equal to $\sigma_z = \rho n_s / \sqrt{n_1 n_2}$ where n_1 is the sample size of the first study, n_2 is the sample size of the second study and ρ is the phenotypic correlation among the n_s overlapping samples. As demonstrated in [9], $\rho =$ $\rho_g + \rho_e$ where ρ_g is the genetic correlation and ρ_e the environmental residual correlation. We generated the residual covariance matrix Σ_r according to two scenarios: (random) ρ_r was randomly sampled with the only constraint that Σ_r remains a semi definite positive matrix, (aligned) ρ_r was constructed as $\rho_r = 0.4 \times \rho_\sigma + 0.6 \times \mathcal{E}$ with \mathcal{E} sampled at random. Z_g varied depending on the simulation scenario: (eff ~ Random) $\rm Z_g$ were sampled from a uniform distribution with boundary [-6; 6], (eff ~ corG) Zg was sampled from a normal multivariate distribution with a random covariance matrix, (eff ~ wG) Zg was sampled along a straight line blurred with a normal noise, (eff ~ high H), Zg was sampled from a normal multivariate distribution simulating genetically uncorrelated traits with high heritability, (eff ~ het H), Zg was samples from a normal multivariate distribution simulating genetically uncorrelated trait with only the first having a high heritability.

Data preprocessing: an overview

The analysis of the 36 GWAS required substantial preprocessing, including the inference of several parameters. First, for many publicly available GWAS, sample size per SNP was not readily available and retrospectively collecting this information can be very challenging as it implies requesting this information from each individual cohort. For such a situation, we propose inferring a proxy for missing sample size as $1/(\hat{\sigma}_{\beta_G}^2 \sigma_G^2)$, where $\hat{\sigma}_{\beta_G}^2$ is the variance of $\hat{\beta}_G$, the estimated SNP effect, and σ_G^2 the variance of the SNP, derived from the coded allele frequency which is either provided with the GWAS or extracted from a reference panel (see **S1 Text**). For linear regression this approximate $N\sigma_e^2$, where *N* is the true sample size and σ_e^2 is a residual variance of the outcome in the regression model. For logistic regression our estimator is a proxy for the term Np(1-p), where *p* is the in-sample proportion of cases, and it therefore assumes that the proportion of cases is relatively stable across SNPs with different sample size.

Another challenging issue was the merging of multiple GWAS with different set of assayed SNPs. Indeed, out of 10 million variants reported for some GWAS, fewer than 1,000 had complete summary statistics for all 36 phenotypes analyzed. We performed an imputation of missing *Z*-scores in each study using the RAISS[27] method we recently developed. The approach uses correlation between SNPs to predict *Z*-score at missing SNPs using available ones and achieves a level of imputation accuracy suitable for multitrait analysis (S1 Text). Here we used the European panels from the 1,000 Genomes project[78] as a reference for the estimation of the *omnibus* statistic (S13 Fig). Nevertheless, as a supplementary quality control (QC), we excluded significant SNPs that were not surrounded by SNPs in linkage disequilibrium with significant or near significant *p*-values ($P < 10^{-6}$).

These two parameter inferences were integrated along other preprocessing operations into a pipeline that is fully described here[28]. Given a reference panel with no ambiguous strand, it consists in the following steps (i) Extract, the coded and alternative alleles, signed statistics (regression coefficient or odds ratio), standard error, p-value, and sample size; (ii) Remove all

SNPs that are not in the reference panel; (iii) Derive *Z*-score for each SNP from signed statistics and *p*-value; (iv) Infer sample size when not available; (v) Remove all SNPs whose sample size is less than 70% of the maximum sample size; and (vi) Infer missing *Z*-scores statistics based on the 1K genome reference panel. After applying our preprocessing pipeline to all 36 GWAS analyzed, there remained 6,978,319 SNPs with a missing rate of 45% (59% before imputation).

Characterization of new loci

To determine new and existing trait-associated loci we used genome regions formed by linkage disequilibrium (LD) blocks as defined in Berisa et al[79] using a reference panel of individuals of European ancestry. It included a total of 1,704 independent regions ranging from 10 kb to 26 Mb in length, with an average size of 1.6 Mb. For each independent LD region, we extracted the minimum *p*-value over all SNPs contained in the region, and a single univariate analysis *p*-value defined as the minimum across all single phenotype GWAS and all SNPs in the region. We consider that a region is newly detected by a multitrait test if the joint analysis *p*-value is genome-wide significant while its univariate p-value is not (joint analysis *p*-value < 1×10^{-8} and univariate *p*-value > 1×10^{-8}). We determined SNPs carrying the signal inside significant region with the plink "clump" function using the following parameters:—clump-p1, 10^{-8} ;— clump-r2, 0.2. We kept the lead SNP by clump for further analysis (gene mapping and clustering).

To report associations exclusively detected in the current report (S4–S10 Tables), we filtered out association present in the GWAS catalogue[1] at the date of the 14th of September 2020 (univariate *p*-value > $5x 10^{-8}$) for traits corresponding to our phenotype set. The following trait labels were used to retrieve associations: (Metabolism set) 'Fasting blood glucose', 'Triglycerides', 'LDL cholesterol', 'LDL cholesterol levels', 'HDL cholesterol levels', 'HDL cholesterol levels', 'Total cholesterol levels', 'HOMA-B', 'HOMA-IR', 'Hemoglobin A1c levels', 'Type 2 diabetes'; (Psychiatric set) 'Schizophrenia', 'Bipolar disorder', 'Major depressive disorder', 'Alzheimer's disease', 'Educational attainment'; (Anthropometry set) 'Height', 'Waist circumference', 'Waist-hip ratio', 'Body mass index', 'Hip circumference'; (Immunity set) 'Bone mineral density', 'Rheumatoid arthritis', 'Ulcerative colitis', 'Inflammatory bowel disease', 'Large artery stroke', 'Stroke', 'Atrial fibrillation', 'Heart rate', 'Heart rate variability traits'; (Composite set) 'Body mass index', 'Waist-hip ratio', 'Triglycerides', 'LDL cholesterol', 'LDL cholesterol levels', 'HDL cholesterol', 'HDL cholesterol', 'LDL cholesterol', 'LDL cholesterol', 'HDL cholesterol', 'HDL cholesterol', 'LDL cholesterol', 'LDL cholesterol', 'HDL cholesterol', 'HDL cholesterol', 'LDL cholesterol', 'LDL cholesterol', 'HDL cholesterol', 'HDL cholesterol', 'LDL cholesterol levels', 'HDL cholesterol', 'HDL cholesterol',

FUN-LDA tissue enrichment

We computed enrichment for SNPs belonging to regions of open chromatin (more likely to contain expressed genes[80, 81]) in specific tissues in three cases: i) when comparing results across phenotype sets, ii) when comparing univariate results, and iii) when comparing results across clusters. For all analyses we used functional annotations on 127 Roadmap tissues and cell lines defined by integrating activating histone marks (H3K4me1, H3K4me3, H3K9ac, and H3K27ac) with a latent Dirichlet allocation model as implemented in FUN-LDA[82]. The enrichment score for a tissue is based on the number significant SNPs compared with the total number of SNPs in open chromatin region (see S1 Text). Enrichment results are reported in S11–S13 Tables.

Multitrait genetic association clustering and selection of the optimal number of clusters

We performed a clustering of top associated SNPs for each phenotype set using a Gaussian Mixture model (GMM). Briefly, the assumed generative model is as follows. Consider n independent vectors $Z_i = (Z_{i1}, \ldots, Z_{ij}, \ldots, Z_{id})^t$ in \mathbb{R}^d , each arising from one of k distinct clusters. Each vector represents one SNPs. The value Z_{ij} is the Z-score of SNP *i* on trait *j*. Let $I_{ij} = 1$ if the *i*th observation belongs to cluster *j*, and define the indicator vector $I_i = (I_{i1}, \ldots, I_{ijj}, \ldots, I_{ik})^t$. Conditional on membership to the *j*th cluster, Z_i follows a multivariate normal distribution, with cluster specific mean μ_j and covariance Σ_j . Let denote π^j the marginal probability of membership to the *j*th cluster. The observations can be viewed as arising from the following hierarchical model:

$$I_i \sim Multinomial(1, \pi)$$

 $Z_i(I_{ij} = 1) = N(\mu_j, \Sigma_j)$

One major difficulty in applying the GMM was to deal with incomplete data. Indeed, even after imputation of some missing statistics, our datasets still contained some missing values. To solve the clustering, we implemented the statistical framework described by Ghahramani et al[83] which we recently implemented in a R package MGMM[40]. This method relies on EM optimization techniques enabling the inference of unobserved variables from observed variables and an assumed Gaussian mixture model. In classical GMM, the only variable inferred is the posterior probability cluster membership. In the MGMM algorithm, missing Z-scores are also inferred taking into account the observed Z-scores and the inferred probability to belong to a cluster.

The model gives for each SNP the posterior probabilities to belong to each cluster, and was therefore assigned to its most likely cluster, as long as its entropy was larger than 0.75. For a given variant *SNP*_i, the entropy was derived as follow:

$$S(SNP_i) = \sum_{j=1}^{k} P(SNP_i \in cluster_j) \times \log(P(SNP_i \in cluster_j))$$

where *k* is the total number of clusters and $P(SNP_i \in cluster_j)$ is the posterior probability of SNP_i to belong to cluster *j*. The higher the entropy the more the SNP attribution to one cluster is ambiguous. SNPs with an entropy higher than 0.75 were filtered out of the clustering results.

Clustering was performed on all independent significant SNPs. For each SNP, we defined three *p*-values on the phenotypic group traits: the minimum univariate *p*-value (P_{univ}), the Sum- Z_{ica} *p*-value and the omnibus *p*-value. All SNPs with at least one of the three *p*-value under 10⁻⁸ were selected for further analysis. For the Metabolism univariate clustering, we only considered the univariate *p*-value to perform the selection. We then applied the plink[84] clump function to retrieve practically independent associations using the 0.2 as clump-r2 parameter and 10⁻⁸ as clump-p1 parameter. For each clump we selected a representative SNPs as the one with the smallest *p*-value across the three tests and having more than 60% of its values observed. Note that for a negligible number of occurrences, the representative SNPs has a p-value above 10⁻⁸ (S15 and S16 Tables). We applied MGMM within each phenotype set and varied the pre-specified number of clusters between 2 and 10. To select the optimal number of clusters *k*, we performed the clustering 100 times on a random subset of 80% of the SNPs for each *k*. For each resulting clustering we computed the Bayesian Information Criteria and the Silhouette[85] (see S23 and S24 Figs). Except for the Metabolism set, the silhouette appears conservative and the

BIC criterion anticonservative, *i.e.* the latter criteria tends to select a larger number of clusters. We decided to use the following *ad hoc* compounded criterion:

- 1. If the optimal number of clusters determined by the BIC criteria is higher than the one determined by the silhouette criteria, starting from the silhouette optimal, increase the number of clusters until one of these two conditions is met: 1) adding one cluster significantly decrease the silhouette criterion, 2) the BIC optimal number is reached.
- 2. In other cases, set the optimal number of clusters to the one determined by silhouette.

Cluster genetic correlation

We defined pairwise genetic covariance per cluster for as $\rho_{g,cluster}=\beta_1^t\beta_2/M$ where β_1 and β_2 are the vector of genetic effects for the pair of phenotypes considered and *M* is the number of SNPs in the cluster. To estimate properly this quantity from the observed $\hat{\beta}$, we accounted for the bias introduced by sample overlap and phenotypic correlation using the following estimator (see S1 Text):

$$\mathbb{E}(\boldsymbol{\beta}_1^T \boldsymbol{\beta}_2) = \frac{\mathbb{E}(\hat{\boldsymbol{\beta}}_1^T \hat{\boldsymbol{\beta}}_2)}{M} - \frac{n_s \rho_Y}{n_1 n_2}$$

where ρ_Y is the phenotypic covariance, and n_s , n_1 and n_2 are respectively the sample size shared between the two traits, for the trait 1, and for the trait 2. To assess whether the estimated genetic covariances are significantly different from zero, we performed for each pair of phenotypes within each cluster, a *t*-test on the vector of random variables (X_1, X_2, \dots, X_M) , were $X_j = \hat{\beta}_{j,1}\hat{\beta}_{j,2} - \frac{n_s\rho}{n_1n_2}$ is the contribution of SNP *j* to the covariance. Note that we used only independent SNPs selected using LD-clumping with squared-correlation parameter equals 0.2.

Functional enrichment of metabolism clusters

We used FUMA[86] *SNP2GENE* function to associate SNPs with genes based on two criteria, the physical position (in 30kb radius of a protein coding gene) and eQTLs (all significant ciseQTL from GTEx up to a distance of 1Mb). Note that we restrained the eQTLs to the one that were found in relevant tissue for the Immunity and Metabolism set: immune cells for Immunity and adipose, intestine, liver and brain tissues for Metabolism (see **S1 Data** for complete parameters). After chaining genes to clusters based on SNPs, we performed a functional enrichment for pathways defined in KEGG[87] and GO[88] databases and derived report *p*-values using FUMA *GENE2FUNC* function. Here, cluster's gene were compared against a background of protein coding genes. Finally, we used the R package pathview[89] to project genes onto KEGG pathways maps.

Disease-clusters association

For the *metabolism* phenotype set, to provide an indicator of the relative contribution of genetic variants to phenotypes in each cluster from the *Metabolism* set, we performed a principal component analysis (PCA) of the SNP-by-phenotype association matrix within each cluster. For this analysis, we used scaled beta coefficients, *i.e. Z*-scores divided by the square root of the phenotype GWAS sample size. To avoid bias due to the arbitrary choice of the coded allele, we randomly shuffled 20 times the coded allele, and repeated the PCA after each shuffling. We report in **Fig.5**, the average of the loadings of the first PC over all shuffling. Note that the first PC only provides the multidimensional direction explaining the largest variance and

therefore do not fully capture the distribution of genetic effect within each cluster. Nevertheless, those first PCs explained a substantial amount of the total variance, equal to 75%, 38%, 53%, 64%, 80% and 93% of the variance in betas for cluster 1 to 6, respectively.

Then, we assessed the association between SNPs within the inferred cluster and three traits (none of which being included in the *Metabolism* set): cardiovascular diseases, any stokes and BMI. SNP alleles were aligned according to the first principal by clusters determined in the last section. We applied a sign test to assess the concordance of the sign of the projection on PC1 and the sign of Z-score for on the three tested additional traits. For this analysis we used more stringent criteria to ensure the SNPs independence. We selected the subset of *Metabolism* SNPs for which linkage disequilibrium does not exceed 0.2 (clump-r2 set to 0.05), which diminishes the number of SNPs considered from 391 to 285. Concerning the association of SNPs to drug target, we associated drug target to a representative SNPs by selecting the SNP with the lowest entropy and having a positive silhouette.

URL Resources

We developed the following R and python packages to perform our study: JASS_Preprocessing: https://gitlab.pasteur.fr/statistical-genetics/jass_preprocessing JASS: https://gitlab.pasteur.fr/statistical-genetics/jass RAISS: https://gitlab.pasteur.fr/statistical-genetics/raiss MGMM: https://cran.r-project.org/web/packages/MGMM/index.html Figs 3–5 have been generated using the following R packages: https://cran.r-project.org/web/packages/UpSetR/index.html https://cran.r-project.org/web/packages/radarchart/README.html https://cran.r-project.org/web/packages/ggalluvial/vignettes/ggalluvial.html https://cran.r-project.org/web/packages/gplots/index.html https://cran.r-project.org/web/packages/gplots/index.html

Supporting information

S1 Text. Supplementary Note. (DOCX)

S1 Fig. Distribution of the proposed statistics under the null. We simulated series of 10,000 replicates, each including z-score statistics for 5 (panels a-c) and 100 (panels d-f) phenotypes and 100K SNPs. For each replicate we applied the *Omnibus* test (panels a and d), the *sumZ* test using weights of 1 (panels c and f) and test *sumZ* test using weights equals to loading of the first principal component of the phenotypic correlation matrix. Left panels show the observed chi-square distribution in blue against the expected one in red. Right panels show the corresponding *p*-value histograms.

(TIF)

S2 Fig. Solving singular covariance matrices. We simulated series of replicates, each including 20,000 individuals, 10,000 SNPs additively coded with frequencies ranging from 0.01 to 0.99, and 100 phenotypes. The phenotypes were drawn from a multivariate normal distribution with mean 0 and a variance-covariance Σ_r of rank 50. For each SNP, we computed the zscore of association with each phenotype using linear regressions and then derived *p*-values from the *Omnibus* test using three strategies to derive the inverse of Σ_r : i) considering only eigenvalues greater than a specified threshold ϵ (left column), ii) replacing eigenvalues below a threshold ϵ by ϵ (middle column), and iii) adding a small value ϵ to the diagonal terms of Σ_r (right column). For each strategy we considered three different ϵ values: 10^{-3} (first line), 10^{-6} (second line), and 10^{-9} (last line). The plots show the histogram of *p*-value distribution. (TIF)

S3 Fig. ICA component selection inflation control. We simulated 1000 z-score vectors under the null hypothesis for each of the 7 GWAS set. The z-score vectors under the null hypothesis follow a multivariate Gaussian of null mean and of covariance given by the intercept of the LDSC regression. The ICA was applied on the 1000 simulated z-scores. For each component, the sumZ test using the component as weight vector. We then selected the component yielding the most association as final component. We computed the p-value for the 1000 point for the optimal component. The y-axis represents the observed p-value quantiles with respect to the theoretical p-value quantiles.



S4 Fig. Summary statistics versus individual-level data tests under the null. We simulated replicates of 10,000 predictors for 500 (left column), 5,000 (middle column) and 10,000 individuals (right column). Predictors were drawn from a binomial with n = 2 and probability varying in [0.01–0.99] to mimic genotype data, and further normalized to have mean 0 and variance 1. We then simulated a) 5, b) 20 and c) 100 correlated outcomes independent from the genotypes. To assess the impact of non-normal outcomes on the relationship between the summary-statistics based test and the individual-level data test, the outcomes were first drawn from a multivariate normal distribution with pairwise correlation ranging in [-0.7; 0.7] and then transform to non-normal distributions based on their quantiles, so that on average, 33% of them follow a uniform distribution, 33% a Laplace distribution, and 33% an exponential distribution. For each SNP, we first applied a Multivariate ANOVA (MANOVA, blue line). We then conducted association screenings for each single phenotype separately, and applied the proposed *Omnibus* test (purple line) and the Wilk's approximation (orange line) on the resulting summary statistics. The plots show the observed -log10(*p*-value) of each test against the expected value under the null.

(TIF)

S5 Fig. Summary statistics versus individual-level data tests under the alternative. We simulated replicates of 10,000 predictors for 500 (left column), 5,000 (middle column) and 10,000 individuals (right column). Predictors were drawn from a binomial with n = 2 and probability varying in [0.01-0.99] to mimic genotype data, and further normalized to have mean 0 and variance 1. We then simulated a) 5, b) 20 and c) 100 correlated outcomes. We assumed that 10% of the predictors were causal, each causal predictor being randomly chosen and associated with up to 50% outcomes, randomly chosen with equal probability. Effect sizes for each causal predictor were drawn from a normal distribution with mean 0 and variance 0.0025 (i.e. the total phenotypic variance explained by the predictors). To assess the impact of non-normal outcomes on the relationship between the summary-statistics based test and the individuallevel data test, the outcomes were first drawn from a multivariate normal distribution with pairwise correlation ranging in [-0.7; 0.7] and then transform to non-normal distributions based on their quantiles, so that on average, 33% of them follow a uniform distribution, 33% a Laplace distribution, and 33% an exponential distribution. For each SNP, we first applied a Multivariate ANOVA (MANOVA, blue line). We then conducted association screenings for each single phenotype separately, and applied the proposed Omnibus test (purple line) and the Wilk's approximation (orange line) on the resulting summary statistics. The plots show the observed -log10(*p*-value) of each test against the expected value under the null. (TIF)

S6 Fig. Validation of the summary-statistics approach in UK Biobank individuals. Multitrait GWAS was performed in real data from the UK Biobank using five traits, 336,347 unrelated individuals, and 14,718 genotyped SNPs on Chromosome 20. We used two methods: MANOVA as implemented in PLINK, and the *Omnibus* test from JASS. The plot shows the -log10(*p*-values) of the Omnibus approach as a function of the -log10(*p*-values) derived with the MANOVA.



S7 Fig. Impact of covariance bias on multivariate tests. We simulated three correlated vectors of z-scores for 10,000 SNPs. We derived the multivariate test for each SNP using either the true covariance matrix (Σ_{true}), an upward-bias covariance matrix (Σ_{upBias} , left panels) or a downward-bias covariance matrix ($\Sigma_{downBias}$, right panels). For both scenarios we derived the *Omnibus* multivariate test using either the true or biased covariance matrix. Invalidity of the test based on the biased covariance matrix is illustrated in the resulting QQplots and genomic inflation factor λ_{GC} (bottom panels).

(TIF)

S8 Fig. Impact of causal variants on the covariance estimation. We simulated series of correlated z-scores for 100,000 SNPs from two outcomes Y_1 and Y_2 . For each simulation we generated a matrix of true genetic effect $\mathbf{b} = (\boldsymbol{\beta}_1 \boldsymbol{\beta}_2)$ of *m* standardized and independent genotypes for two phenotypes Y_1 and Y_2 from a multivariate normal with means of 0 variance h_1^2/m , and h_2^2/m , respectively, and covariance σ_{g_2} where $h_1^2 = 0.3$ and $h_2^2 = 0.6$ are the heritability of Y_1 and Y_2 . We then generated $\hat{\mathbf{b}}$ defined as $\hat{\mathbf{b}} = \mathbf{b} + \boldsymbol{\varepsilon}$ where $\boldsymbol{\varepsilon}$ was also drawn from a multivariate normal with means 0, variance $1/N_1$ and $1/N_2$, respectively, and covariance $r_e N_s / \sqrt{N_1 N_2}$, where N_1 , $N_2 = N_1/2$ and $N_s = N_2/2$ are the sample sizes for Y_1 and Y_2 , and the number of shared samples, respectively, and $r_e = 0.4$ is the correlation between Y_1 and Y_2 across overlapping samples. Finally, we derived the expected z-score for each genotype $\mathbf{z} = (\hat{\boldsymbol{\beta}}_1 \sqrt{N_1} \ \hat{\boldsymbol{\beta}}_2 \sqrt{N_2}) = (\mathbf{z}_1 \ \mathbf{z}_2)$, and $\boldsymbol{\sigma}_z = \boldsymbol{cov}(\mathbf{z}_1, \mathbf{z}_2)$, the covariance between \mathbf{z}_1 and \mathbf{z}_2 . The left panel (a) show the covariance between z-score for null variants in red, and the observed covariance between all z-scores except those harboring a p-value below a given threshold (0 (i.e. no SNP removed), $5x10^{-8}$, $5x10^{-5}$, $5x10^{-4}$, $5x10^{-3}$, $1x10^{-2}$, and $5x10^{-8}$). The right panel (b) shows the same results while assuming only 10% of the variants are causal, while all remaining have **b** = **0**.

(TIF)

S9 Fig. Validation of the LDSC covariance estimate accuracy in UK Biobank. We compared LDSC estimates of covariance between summary statistics against its expected value, which equals $\rho N_s / \sqrt{N_1 N_2}$, where ρ is the phenotypic correlation among overlapping sample and N_s , N_1 and N_2 are the sample overlap, the sample size for phenotype 1 and the sample size for phenotype 2, respectively. We used individual-level data for five anthropometric traits and 619,017 SNPS measured in 336,347 individuals from the UK Biobank cohort. We first considered scenarios with complete sample overlap (i.e. $N_s = N_1 = N_2$, panels a-d), so that LDSC estimate is expected to equal the phenotypic correlation. We compared estimates while using 100% (a), 50% (b), 10% (c) and 1% (ds) of the total sample size respectively. The lower matrix triangle in pale red are estimated phenotypic correlation derived using the LDSC. We then considered scenario where sample overlap is only partial by sub-sampling individuals only for BMI, using 100% (e),50% (f), 10% (g) and 1% (h) of the total sample for that phenotype. The first row, in red, is the expected GWAS covariance knowing ρ , N_s , N_1 and N_2 . The second row,

in blue, is the estimate derived using LDSC. (TIF)

S10 Fig. Bias due to sample size heterogeneity in the GLG consortium GWAS data. We performed the *Omnibus* test to the four traits from the GLG consortium: high density lipoprotein (HDL), low density lipoprotein (LDL), total cholesterol (TC), and total triglyceride (TG) using all available SNP with complete summary statistics. We plotted the median chi-squared of the resulting test across bin of SNPs defined based the per-SNP standard deviation in sample size across the four traits ($\sigma_{sample size}$) for SNPs with a minor allele frequency (MAF) below 5% (a), and above 5% (b). The red dashed line indicates the expected 4 degree of freedom chi-square under the null. The shade of blue is proportional to the number of SNPs in each bin. (TIF)

S11 Fig. Impact of allele frequency error on sample size inference. We simulated 1,000 replicates including 20,000 individuals were a phenotype Y is simulated independently of a genotype G with frequency randomly sampled in [0.001, 0.999]. For each replicate we tested for association between G and Y and we inferred w the weights, which equals the sample size times a constant, using the standard error of the effect estimate and either the in-sample allele frequency or the in-sample allele plus a noise term sample from a uniform with min and max in [0.01, 0.05, 0.10]. When allele frequency plus noise was larger or smaller than 0 or 1, the value was set to 0.001 and 0.999, respectively. Upper and lower panels show the inferred sample size as a function of the true allele frequency when using the identity and logit link functions for modelling and testing for association, respectively. Note that for the later, for each replicate, we simulated 50,000 individuals and considered a disease prevalence of 25%. We then randomly sampled 10,000 cases and 10,000 controls to form replicates of 20,000 individuals. Also, for the sake of comparison, we scaled w by a constant in the logit model so that the target is the true sample size.

(TIF)

S12 Fig. Impact of case/control ratio misspecification on sample size inference. We simulated 1,000 replicates including 50,000 individuals. For each individual we generated a disease status assuming a prevalence of 25% and an independent genotype **G** with frequency randomly sampled in [0.001, 0.999]. For each replicate we randomly sampled 5,000 cases and 20,000 controls and tested for association between **G** and **Y** using these samples and after sub-sampling cases (i.e. using a sub-sample of the 5,000 cases and the 20,000 controls, top panel) or controls (i.e. using the 5,000 cases and a sub-sample of the 20,000 controls, bottom panel), in order to mimic situation where either cases or controls would be missing for some SNPs. For each experiment we inferred $W_{log} = Np(1-p)$, where N is the sample size and **p** is the case-control ratio, using the standard error of the effect estimate and the in-sample allele frequency. In both situations (sub-sampling cases or sub-sampling controls), W_{log} is decreasing with increasing percentage of missingness.

(TIF)

S13 Fig. Effect of imputation of missing z-scores on the Omnibus test statistic. Each line corresponds to a GWAS set. The left column is the empirical quantile of the *Omnibus* statistic versus the theoretical quantile before imputation. The middle column is the same after imputation. The right column is the empirical quantile of the *Omnibus* statistic before imputation versus the same quantity after imputation. (TIF)

PLOS Genetics | https://doi.org/10.1371/journal.pgen.1009713 August 30, 2021

S14 Fig. Bivariate Significance boundary for each test. We simulated series of N = 100K individuals with two correlated outcomes Y_1 and Y_2 as a function of m = 1000 independent SNPs. The genetic effect of the m variants on Y_1 and Y_1 , denoted $\mathbf{b} = (\beta_1 \beta_2)$, were drawn from a bivariate normal with means 0, variance h_1^2/m , and h_2^2/m , and correlation σ_g , where $h_1^2 = 0.4$, $h_2^2 = 0.25$, and $\sigma_g = 0.8$. The vectors of residual $\boldsymbol{\varepsilon} = (\boldsymbol{\varepsilon}_1, \boldsymbol{\varepsilon}_2)$ were drawn from a multivariate normal with means 0, variance $1 - h_1^2$ and $1 - h_2^2$, respectively, and covariance $r_e = 0.5$. For each simulation, we derived the z-score for each of the m genotypes $\mathbf{z} = (\hat{\beta}_1 \sqrt{N} \quad \hat{\beta}_2 \sqrt{N}) = (z_1 \quad z_2)$ and plotted z_1 as a function of z_2 . We applied the five tests: *Omnibus*, $sumZ_r$, $sumZ_g$, $sumZ_{ica}$ and $sumZ_I$ and highlighted for each of them whether variants had significant p-value (red, $p < 5x10^{-5}$) or not (blue, $p > 5x10^{-5}$). In the results from the upper panels, \mathbf{b} was drawn from a mixture of two normal distributions both centered on zero and both with variance equal to h_1^2/m , and h_2^2/m . However, the covariance σ_g was set to 0.8 for the first one and -0.65 for the second one. (TIF)

S15 Fig. Signal comparison for all phenotypes. The upper panel shows independent variants detected across phenotype groups and across approaches represented as an *UpSetR* visualization. Matrix lines correspond to a test, each column to a set of significant variants. For each set, the test for which variants are significant are represented with a black dot on the test line. The barplot on the left of the matrix represents the number of significant independent signals detected by each approach. The barplot on the top of the matrix represents the cardinality of the sets. The sets are ordered by cardinality from the largest to the leftmost to the smallest to the rightmost. The bottom panels show quadrant plots, i.e. the -log10(*p*-value) for the most significant SNP per region for the *Omnibus* test as a function of the -log10(*p*-value) for the most significant SNP per region across all univariate GWAS. Complete results are presented in the left panel, and a zoom around the genome-wide significance threshold is presented on the right panel.

(TIF)

S16 Fig. Signal comparison for anthropometric traits. The upper panel shows independent variants detected across phenotype groups and across approaches represented as an *UpSetR* visualization. Matrix lines correspond to a test, each column to a set of significant variants. For each set, the test for which variants are significant are represented with a black dot on the test line. The barplot on the left of the matrix represents the number of significant independent signals detected by each approach. The barplot on the top of the matrix represents the cardinality of the sets. The sets are ordered by cardinality from the largest to the leftmost to the smallest to the rightmost. The bottom panels show quadrant plots, i.e. the -log10(*p*-value) for the most significant SNP per region for the *Omnibus* test as a function of the -log10(*p*-value) for the most significant SNP per region across all univariate GWAS. Complete results are presented in the left panel, and a zoom around the genome-wide significance threshold is presented on the right panel.

(TIF)

S17 Fig. Signal comparison for cardiovascular phenotypes. The upper panel shows independent variants detected across phenotype groups and across approaches represented as an *UpSetR* visualization. Matrix lines correspond to a test, each column to a set of significant variants. For each set, the test for which variants are significant are represented with a black dot on the test line. The barplot on the left of the matrix represents the number of significant

independent signals detected by each approach. The barplot on the top of the matrix represents the cardinality of the sets. The sets are ordered by cardinality from the largest to the leftmost to the smallest to the rightmost. The bottom panels show quadrant plots, i.e. the -log10 (*p*-value) for the most significant SNP per region for the *Omnibus* test as a function of the -log10(*p*-value) for the most significant SNP per region across all univariate GWAS. Complete results are presented in the left panel, and a zoom around the genome-wide significance threshold is presented on the right panel. (TIF)

S18 Fig. Signal comparison for composite phenotypes set. The upper panel shows independent variants detected across phenotype groups and across approaches represented as an *UpSetR* visualization. Matrix lines correspond to a test, each column to a set of significant variants. For each set, the test for which variants are significant are represented with a black dot on the test line. The barplot on the left of the matrix represents the number of significant independent signals detected by each approach. The barplot on the top of the matrix represents the cardinality of the sets. The sets are ordered by cardinality from the largest to the leftmost to the smallest to the rightmost. The bottom panels show quadrant plots, i.e. the -log10(*p*-value) for the most significant SNP per region across all univariate GWAS. Complete results are presented in the left panel, and a zoom around the genome-wide significance threshold is presented on the right panel.

(TIF)

S19 Fig. Signal comparison for immunity phenotypes. The upper panel shows independent variants detected across phenotype groups and across approaches represented as an *UpSetR* visualization. Matrix lines correspond to a test, each column to a set of significant variants. For each set, the test for which variants are significant are represented with a black dot on the test line. The barplot on the left of the matrix represents the number of significant independent signals detected by each approach. The barplot on the top of the matrix represents the cardinality of the sets. The sets are ordered by cardinality from the largest to the leftmost to the smallest to the rightmost. The bottom panels show quadrant plots, i.e. the -log10(*p*-value) for the most significant SNP per region for the *Omnibus* test as a function of the -log10(*p*-value) for the most significant SNP per region across all univariate GWAS. Complete results are presented in the left panel, and a zoom around the genome-wide significance threshold is presented on the right panel.



S20 Fig. Signal comparison for metabolism phenotypes. The upper panel shows independent variants detected across phenotype groups and across approaches represented as an *UpSetR* visualization. Matrix lines correspond to a test, each column to a set of significant variants. For each set, the test for which variants are significant are represented with a black dot on the test line. The barplot on the left of the matrix represents the number of significant independent signals detected by each approach. The barplot on the top of the matrix represents the cardinality of the sets. The sets are ordered by cardinality from the largest to the leftmost to the smallest to the rightmost. The bottom panels show quadrant plots, i.e. the -log10(*p*-value) for the most significant SNP per region for the *Omnibus* test as a function of the -log10(*p*-value) for the most significant SNP per region across all univariate GWAS. Complete results are presented in the left panel, and a zoom around the genome-wide significance threshold is presented on the right panel.

(TIF)

S21 Fig. Signal comparison for psychiatric phenotypes. The upper panel shows independent variants detected across phenotype groups and across approaches represented as an *UpSetR* visualization. Matrix lines correspond to a test, each column to a set of significant variants. For each set, the test for which variants are significant are represented with a black dot on the test line. The barplot on the left of the matrix represents the number of significant independent signals detected by each approach. The barplot on the top of the matrix represents the cardinality of the sets. The sets are ordered by cardinality from the largest to the leftmost to the smallest to the rightmost. The bottom panels show quadrant plots, i.e. the -log10(*p*-value) for the most significant SNP per region for the *Omnibus* test as a function of the -log10(*p*-value) for the most significant SNP per region across all univariate GWAS. Complete results are presented in the left panel, and a zoom around the genome-wide significance threshold is presented on the right panel.



S22 Fig. Proportion of tissue type enriched by phenotype group. The union of enriched tissue by phenotype set were mapped to their larger anatomical category (Tissue type) in GTEx. To simplify the visualization, anatomical categories that did not explain at least 5% of any of the phenotype group were regrouped under the "Other" category. In each group, anatomical category representing less than 1% were approximated to 0. a) results obtained with variants detected with univariate test and b) results with variants detected with multivariate tests. (TIF)

S23 Fig. Clustering criterion by number of clusters and phenotype set. In all panels, the xaxis is the number of clusters derived using the union of significant SNPs in the *Omnibus*, $SumZ_{genet}$ and univariate tests. On the left column, the y-axis represents the BIC (Bayesian Information Criteria). On the right column, the y-axis is the Silhouette criteria (see Methods). Each line corresponds to a different group of phenotypes. (TIF)

S24 Fig. Clustering for the *Metabolism* **set using SNPs detected by univariate analysis only.** The x-axis is the number of clusters derived using the significant SNPs univariate tests. On the left column, the y-axis represents the BIC (Bayesian Information Criteria). On the right column, the y-axis is the Silhouette criteria (see Methods). (TIF)

S25 Fig. Alluvial plot and heatmaps for the Anthropometry GWAS set. On the left panel, the alluvial plot represents the re-assignment of SNPs from univariate analysis to clusters. To emphasize the relative genetic contribution to phenotypes, SNPs from each phenotype block were weighted by their variance explained to that phenotype. On the right panel, the heatmap represents the multi-trait signatures. Each line is a SNPs, each column, a trait. The gradient of color represents the strength of the Z-scores. (TIF)

S26 Fig. Alluvial plot and heatmap for the Cardiovascular GWAS set. On the left panel, the alluvial plot represents the re-assignment of SNPs from univariate analysis to clusters. To emphasize the relative genetic contribution to phenotypes, SNPs from each phenotype block were weighted by their variance explained to that phenotype. On the right panel, the heatmap represents the multi-trait signatures. Each line is a SNP, each column, a trait. The gradient of color represents the strength of the Z-scores. (TIF)

S27 Fig. Alluvial plot and heatmap for the Immunity GWAS set. On the left panel, the alluvial plot represents the re-assignment of SNPs from univariate analysis to clusters. To emphasize the relative genetic contribution to phenotypes, SNPs from each phenotype block were weighted by their variance explained to that phenotype. On the right panel, the heatmap represents the multi-trait signatures. Each line is a SNP, each column, a trait. The gradient of color represents the strength of the Z-scores. (TIF)

S28 Fig. Alluvial plot and heatmap for the Composite GWAS set. On the left panel, the alluvial plot represents the re-assignment of SNPs from univariate analysis to clusters. To emphasize the relative genetic contribution to phenotypes, SNPs from each phenotype block were weighted by their variance explained to that phenotype. On the right panel, the heatmap represents the multi-trait signatures. Each line is a SNP, each column, a trait. The gradient of color represents the strength of the Z-scores. (TIF)

S29 Fig. Alluvial plot and heatmap for the Psychiatric GWAS set. On the left panel, the alluvial plot represents the re-assignment of SNPs from univariate analysis to clusters. To emphasize the relative genetic contribution to phenotypes, SNPs from each phenotype block were weighted by their variance explained to that phenotype. On the right panel, the heatmap represents the multi-trait signatures. Each line is a SNP, each column, a trait. The gradient of color represents the strength of the Z-scores. (TIF)

S30 Fig. Alluvial plot and heatmap for the Metabolism GWAS set. On the left panel, the alluvial plot represents the re-assignment of SNPs from univariate analysis to clusters. To emphasize the relative genetic contribution to phenotypes, SNPs from each phenotype block were weighted by their variance explained to that phenotype. On the right panel, the heatmap represents the multi-trait signatures. Each line is a SNP, each column, a trait. The gradient of color represents the strength of the Z-scores. (TIF)

S31 Fig. Alluvial plot and heatmap for all GWAS combined. On the left panel, the alluvial plot represents the re-assignment of SNPs from univariate analysis to clusters. To emphasize the relative genetic contribution to phenotypes, SNPs from each phenotype block were weighted by their variance explained to that phenotype. On the right panel, the heatmap represents the multi-trait signatures. Each line is a SNP, each column, a trait. The gradient of color represents the strength of the Z-scores. (TIF)

S32 Fig. Heatmap for the *Metabolism* **SNPs for cardiovascular traits and Alzheimer disease.** the heatmap represents the impact multi-trait signatures detected on the METABOLISM set on disease that have been linked to hyperlipidemia and diabetes. Each line is a SNP, each column, a trait. The gradient of color represents the strength of the Z-scores. (TIF)

S1 Table. GWAS considered.

(XLSX)

S2 Table. Variance-covariance matrix of Z-score under the null hypothesis. (XLSX)

S3 Table. Genetic correlation. (XLSX) S4 Table. New associations for the Metabolism phenotype set. (XLSX) S5 Table. New associations for the Cardiovascular phenotype set. (XLSX) S6 Table. New associations for the Immunity phenotype set. (XLSX) S7 Table. New associations for the Anthropometry phenotype set. (XLSX) S8 Table. New associations for the Psychiatric phenotype set. (XLSX) S9 Table. New associations for the Composite phenotype set. (XLSX) S10 Table. New associations for the all the phenotypes considered together. (XLSX) S11 Table. Significant tissue enrichment by joint test and by phenotype set. (XLSX) S12 Table. Significant tissue enrichment for univariate GWAS derived with FUN-LDA. (XLSX) S13 Table. Significant tissue enrichment per multitrait association profile derived with FUN-LDA. (XLSX) S14 Table. Significant Go term enrichment for all phenotype sets and multitrait association profiles with FUMA. (XLSX) S15 Table. Significant KEGG term enrichment by phenotype sets and multitrait association profiles with FUMA. (XLSX) S16 Table. Significant SNPs for the Metabolism set that were retained in final clusters. (XLSX) S17 Table. Significant SNPs for the Immunity set that were retained in final clusters. (XLSX) S18 Table. Significant SNPs for the METABOLISM set retained in final clusters and with negligible linkage disequilibrium (R2 < 0.05). (XLSX) S19 Table. Sign test for z-score of diseases phenotype associated with metabolic traits. (XLSX)

S20 Table. Potential drug targets mapped to multi-trait profiles for the Metabolism, Immunity, Cardiovascular phenotype set. (XLSX)

S1 Data. FUMA parameter for functional analysis.

(ZIP)

Acknowledgments

This research has been conducted using the UK Biobank Resource under Application Number 42260.

Author Contributions

Conceptualization: Hanna Julienne, Bjarni J. Vilhjálmsson, Hugues Aschard.

Data curation: Hanna Julienne, Amaury Vaysse.

- Formal analysis: Hanna Julienne, Vincent Laville, Zihuai He, Carla Lasry, Andrey Ziyatdinov, Bjarni J. Vilhjálmsson, Hugues Aschard.
- Methodology: Hanna Julienne.
- Software: Hanna Julienne, Zachary R. McCaw, Vincent Guillemot, Cyril Nerin, Pierre Lechat, Hervé Ménager.

Supervision: Hugues Aschard.

Writing - original draft: Hanna Julienne, Hugues Aschard.

Writing – review & editing: Vincent Laville, Zachary R. McCaw, Wilfried Le Goff, Marie-Pierre Dube, Peter Kraft, Iuliana Ionita-Laza, Bjarni J. Vilhjálmsson, Hugues Aschard.

References

- Buniello A, MacArthur JAL, Cerezo M, Harris LW, Hayhurst J, Malangone C, et al. The NHGRI-EBI GWAS Catalog of published genome-wide association studies, targeted arrays and summary statistics 2019. Nucleic acids research. 2019; 47(D1):D1005–D12. Epub 2018/11/18. https://doi.org/10.1093/nar/ gky1120 PMID: 30445434; PubMed Central PMCID: PMC6323933.
- 2. Pasaniuc B, Price AL. Dissecting the genetics of complex traits using summary association statistics. Nature Reviews Genetics. 2016; 18(2):117–27. https://doi.org/10.1038/nrg.2016.142 PMID: 27840428
- Bulik-Sullivan BK, Loh PR, Finucane HK, Ripke S, Yang J, Schizophrenia Working Group of the Psychiatric Genomics C, et al. LD Score regression distinguishes confounding from polygenicity in genomewide association studies. Nat Genet. 2015; 47(3):291–5. Epub 2015/02/03. https://doi.org/10.1038/ng. 3211 PMID: 25642630; PubMed Central PMCID: PMC4495769.
- Euesden J, Lewis CM, O'Reilly PF. PRSice: Polygenic Risk Score software. Bioinformatics. 2015; 31 (9):1466–8. Epub 2015/01/01. https://doi.org/10.1093/bioinformatics/btu848 PMID: 25550326; PubMed Central PMCID: PMC4410663.
- Do R, Willer CJ, Schmidt EM, Sengupta S, Gao C, Peloso GM, et al. Common variants associated with plasma triglycerides and risk for coronary artery disease. Nature genetics. 2013; 45(11):1345–52. Epub 2013/10/08. https://doi.org/10.1038/ng.2795 PMID: 24097064; PubMed Central PMCID: PMC3904346.
- Pickrell JK, Berisa T, Liu JZ, Ségurel L, Tung JY, Hinds DA. Detection and interpretation of shared genetic influences on 42 human traits. Nature genetics. 2016; 48(7):709–17. <u>https://doi.org/10.1038/ng. 3570 PMID: 27182965</u>
- Watanabe K, Stringer S, Frei O, Umicevic Mirkov M, de Leeuw C, Polderman TJC, et al. A global overview of pleiotropy and genetic architecture in complex traits. Nat Genet. 2019; 51(9):1339–48. Epub 2019/08/21. https://doi.org/10.1038/s41588-019-0481-0 PMID: 31427789.
- van Rheenen W, Peyrot WJ, Schork AJ, Lee SH, Wray NR. Genetic correlations of polygenic disease traits: from theory to practice. Nat Rev Genet. 2019. Epub 2019/06/07. https://doi.org/10.1038/s41576-019-0137-z PMID: 31171865.

- Bulik-Sullivan B, Finucane HK, Anttila V, Gusev A, Day FR, Loh PR, et al. An atlas of genetic correlations across human diseases and traits. Nat Genet. 2015; 47(11):1236–41. Epub 2015/09/29. https://doi.org/10.1038/ng.3406 PMID: 26414676; PubMed Central PMCID: PMC4797329.
- Shi H, Mancuso N, Spendlove S, Pasaniuc B. Local Genetic Correlation Gives Insights into the Shared Genetic Architecture of Complex Traits. American journal of human genetics. 2017; 101(5):737–51. Epub 2017/11/04. https://doi.org/10.1016/j.ajhg.2017.09.022 PMID: 29100087; PubMed Central PMCID: PMC5673668.
- Cichonska A, Rousu J, Marttinen P, Kangas AJ, Soininen P, Lehtimaki T, et al. metaCCA: summary statistics-based multivariate meta-analysis of genome-wide association studies using canonical correlation analysis. Bioinformatics. 2016; 32(13):1981–9. Epub 2016/05/07. https://doi.org/10.1093/bioinformatics/btw052 PMID: 27153689; PubMed Central PMCID: PMC4920109.
- Qi G, Chatterjee N. Heritability informed power optimization (HIPO) leads to enhanced detection of genetic associations across multiple traits. PLoS Genet. 2018; 14(10):e1007549. Epub 2018/10/06. https://doi.org/10.1371/journal.pgen.1007549 PMID: 30289880.
- 13. Wang Z, Sha Q, Zhang S. Joint Analysis of Multiple Traits Using "Optimal" Maximum Heritability Test. PLoS One. 2016; 11(3):e0150975. Epub 2016/03/08. https://doi.org/10.1371/journal.pone.0150975 PMID: 26950849; PubMed Central PMCID: PMC4780705.
- Zhu X, Feng T, Tayo BO, Liang J, Young JH, Franceschini N, et al. Meta-analysis of correlated traits via summary statistics from GWASs with an application in hypertension. American Journal of Human Genetics. 2015; 96(1):21–36. https://doi.org/10.1016/j.ajhg.2014.11.011 PMID: 25500260
- Kim J, Bai Y, Pan W. An Adaptive Association Test for Multiple Phenotypes with GWAS Summary Statistics. Genet Epidemiol. 2015; 39(8):651–63. Epub 2015/10/24. <u>https://doi.org/10.1002/gepi.21931</u> PMID: 26493956; PubMed Central PMCID: PMC4715495.
- Province MA, Borecki IB. A correlated meta-analysis strategy for data mining "OMIC" scans. Pac Symp Biocomput. 2013:236–46. Epub 2013/02/21. PMID: <u>23424128</u>; PubMed Central PMCID: PMC3773990.
- Guo B, Wu B. Integrate multiple traits to detect novel trait-gene association using GWAS summary data with an adaptive test approach. Bioinformatics. 2018. Epub 2018/11/27. https://doi.org/10.1093/ bioinformatics/bty961 PMID: 30476000; PubMed Central PMCID: PMC6596889.
- Liu Z, Lin X. Multiple phenotype association tests using summary statistics in genome-wide association studies. Biometrics. 2018; 74:165–75. https://doi.org/10.1111/biom.12735 PMID: 28653391
- Korte A, Vilhjalmsson BJ, Segura V, Platt A, Long Q, Nordborg M. A mixed-model approach for genome-wide association studies of correlated traits in structured populations. Nat Genet. 2012; 44 (9):1066–71. Epub 2012/08/21. <u>https://doi.org/10.1038/ng.2376</u> PMID: <u>22902788</u>; PubMed Central PMCID: PMC3432668.
- Solovieff N, Cotsapas C, Lee PH, Purcell SM, Smoller JW. Pleiotropy in complex traits: Challenges and strategies. Nature Reviews Genetics. 2013; 14(7):483–95. https://doi.org/10.1038/nrg3461 PMID: 23752797
- van der Meer D, Frei O, Kaufmann T, Shadrin AA, Devor A, Smeland OB, et al. Understanding the genetic determinants of the brain with MOSTest. Nat Commun. 2020; 11(1):3512. Epub 2020/07/16. <u>https://doi.org/10.1038/s41467-020-17368-1</u> PMID: <u>32665545</u>; PubMed Central PMCID: PMC7360598.
- Pegolo S, Momen M, Morota G, Rosa GJM, Gianola D, Bittante G, et al. Structural equation modeling for investigating multi-trait genetic architecture of udder health in dairy cattle. Sci Rep. 2020; 10 (1):7751. Epub 2020/05/10. https://doi.org/10.1038/s41598-020-64575-3 PMID: 32385377; PubMed Central PMCID: PMC7210309.
- Baselmans BML, Jansen R, Ip HF, van Dongen J, Abdellaoui A, van de Weijer MP, et al. Multivariate genome-wide analyses of the well-being spectrum. Nat Genet. 2019; 51(3):445–51. Epub 2019/01/16. https://doi.org/10.1038/s41588-018-0320-8 PMID: 30643256.
- Zhou X, Stephens M. Efficient multivariate linear mixed model algorithms for genome-wide association studies. Nat Methods. 2014; 11(4):407–9. Epub 2014/02/18. https://doi.org/10.1038/nmeth.2848 PMID: 24531419; PubMed Central PMCID: PMC4211878.
- Aschard H, Vilhjalmsson BJ, Greliche N, Morange PE, Tregouet DA, Kraft P. Maximizing the power of principal-component analysis of correlated phenotypes in genome-wide association studies. Am J Hum Genet. 2014; 94(5):662–76. Epub 2014/04/22. https://doi.org/10.1016/j.ajhg.2014.03.016 PMID: 24746957; PubMed Central PMCID: PMC4067564.
- Turley P, Walters RK, Maghzian O, Okbay A, Lee JJ, Fontana MA, et al. Multi-trait analysis of genomewide association summary statistics using MTAG. Nat Genet. 2018; 50(2):229–37. Epub 2018/01/03. https://doi.org/10.1038/s41588-017-0009-4 PMID: 29292387; PubMed Central PMCID: PMC5805593.

- Julienne H, Shi H, Pasaniuc B, Aschard H. RAISS: Robust and Accurate imputation from Summary Statistics. Bioinformatics. 2019. Epub 2019/06/08. https://doi.org/10.1093/bioinformatics/btz466 PMID: 31173064.
- Julienne H, Lechat P, Guillemot V, Lasry C, Yao C, Araud R, et al. JASS: command line and web interface for the joint analysis of GWAS results. NAR Genom Bioinform. 2020; 2(1):lqaa003. Epub 2020/02/ 01. https://doi.org/10.1093/nargab/lgaa003 PMID: 32002517; PubMed Central PMCID: PMC6978790.
- 29. van der Sluis S, Posthuma D, Dolan CV. TATES: efficient multivariate genotype-phenotype analysis for genome-wide association studies. PLoS genetics. 2013; 9(1):e1003235. Epub 2013/01/30. https://doi.org/10.1371/journal.pgen.1003235 PMID: 23359524; PubMed Central PMCID: PMC3554627.
- Ray D, Boehnke M. Methods for meta-analysis of multiple traits using GWAS summary statistics. Genetic epidemiology. 2018; 42(2):134–45. Epub 2017/12/12. https://doi.org/10.1002/gepi.22105 PMID: 29226385; PubMed Central PMCID: PMC5811402.
- Udler MS, Kim J, von Grotthuss M, Bonas-Guarch S, Cole JB, Chiou J, et al. Type 2 diabetes genetic loci informed by multi-trait associations point to disease mechanisms and subtypes: A soft clustering analysis. PLoS Med. 2018; 15(9):e1002654. Epub 2018/09/22. https://doi.org/10.1371/journal.pmed. 1002654 PMID: 30240442; PubMed Central PMCID: PMC6150463.
- Tanigawa Y, Li J, Justesen JM, Horn H, Aguirre M, DeBoever C, et al. Components of genetic associations across 2,138 phenotypes in the UK Biobank highlight adipocyte biology. Nat Commun. 2019; 10 (1):4064. Epub 2019/09/08. https://doi.org/10.1038/s41467-019-11953-9 PMID: 31492854; PubMed Central PMCID: PMC6731283.
- Cotsapas C, Voight BF, Rossin E, Lage K, Neale BM, Wallace C, et al. Pervasive sharing of genetic effects in autoimmune disease. PLoS genetics. 2011; 7(8):e1002254. Epub 2011/08/20. https://doi.org/ 10.1371/journal.pgen.1002254 PMID: 21852963; PubMed Central PMCID: PMC3154137.
- Dong H, Zhou W, Wang P, Zuo E, Ying X, Chai S, et al. Comprehensive Analysis of the Genetic and Epigenetic Mechanisms of Osteoporosis and Bone Mineral Density. Front Cell Dev Biol. 2020; 8:194. Epub 2020/04/10. <u>https://doi.org/10.3389/fcell.2020.00194</u> PMID: <u>32269995</u>; PubMed Central PMCID: PMC7109267.
- Liu X, Zhang Y, Tian J, Gao F. Analyzing Genome-Wide Association Study Dataset Highlights Immune Pathways in Lip Bone Mineral Density. Front Genet. 2020; 11:4. Epub 2020/03/27. https://doi.org/10. 3389/fgene.2020.00004 PubMed Central PMCID: PMC7077504. PMID: 32211016
- Andreassen OA, Desikan RS, Wang Y, Thompson WK, Schork AJ, Zuber V, et al. Abundant genetic overlap between blood lipids and immune-mediated diseases indicates shared molecular genetic mechanisms. PloS one. 2015; 10(4):e0123057. Epub 2015/04/09. https://doi.org/10.1371/journal.pone. 0123057 PMID: 25853426; PubMed Central PMCID: PMC4390360.
- Michel M, Schmidt MJ, Mirnics K. Immune system gene dysregulation in autism and schizophrenia. Dev Neurobiol. 2012; 72(10):1277–87. Epub 2012/07/04. <u>https://doi.org/10.1002/dneu.22044</u> PMID: 22753382; PubMed Central PMCID: PMC3435446.
- Berthoud HR, Munzberg H, Morrison CD. Blaming the Brain for Obesity: Integration of Hedonic and Homeostatic Mechanisms. Gastroenterology. 2017; 152(7):1728–38. Epub 2017/02/14. https://doi.org/ 10.1053/j.gastro.2016.12.050 PMID: 28192106; PubMed Central PMCID: PMC5406238.
- Roh E, Song DK, Kim MS. Emerging role of the brain in the homeostatic regulation of energy and glucose metabolism. Exp Mol Med. 2016; 48:e216. Epub 2016/03/12. https://doi.org/10.1038/emm.2016.4 PMID: 26964832; PubMed Central PMCID: PMC4892882.
- McCaw ZR, Julienne H, Aschard H. MGMM: an R package for fitting Gaussian Mixture Models on Incomplete Genomics Data. bioRxiv. 2019. https://doi.org/10.1101/2019.12.20.884551.
- Rosenson RS, Davidson MH, Hirsh BJ, Kathiresan S, Gaudet D. Genetics and causality of triglyceriderich lipoproteins in atherosclerotic cardiovascular disease. J Am Coll Cardiol. 2014; 64(23):2525–40. Epub 2014/12/17. https://doi.org/10.1016/j.jacc.2014.09.042 PMID: 25500239.
- Tall AR, Yvan-Charvet L. Cholesterol, inflammation and innate immunity. Nat Rev Immunol. 2015; 15 (2):104–16. Epub 2015/01/24. <u>https://doi.org/10.1038/nri3793</u> PMID: <u>25614320</u>; PubMed Central PMCID: PMC4669071.
- Grebe A, Hoss F, Latz E. NLRP3 Inflammasome and the IL-1 Pathway in Atherosclerosis. Circulation research. 2018; 122(12):1722–40. Epub 2018/06/09. https://doi.org/10.1161/CIRCRESAHA.118. 311362 PMID: 29880500.
- 44. Catapano AL, Pirillo A, Bonacina F, Norata GD. HDL in innate and adaptive immunity. Cardiovasc Res. 2014; 103(3):372–83. Epub 2014/06/18. https://doi.org/10.1093/cvr/cvu150 PMID: 24935428.
- Madsen CM, Varbo A, Tybjaerg-Hansen A, Frikke-Schmidt R, Nordestgaard BG. U-shaped relationship of HDL and risk of infectious disease: two prospective population-based cohort studies. Eur Heart J. 2018; 39(14):1181–90. Epub 2017/12/12. https://doi.org/10.1093/eurheartj/ehx665 PMID: 29228167.

- 46. Ference BA, Ginsberg HN, Graham I, Ray KK, Packard CJ, Bruckert E, et al. Low-density lipoproteins cause atherosclerotic cardiovascular disease. 1. Evidence from genetic, epidemiologic, and clinical studies. A consensus statement from the European Atherosclerosis Society Consensus Panel. Eur Heart J. 2017; 38(32):2459–72. Epub 2017/04/27. https://doi.org/10.1093/eurheartj/ehx144 PMID: 28444290; PubMed Central PMCID: PMC5837225.
- Emerging Risk Factors C, Di Angelantonio E, Gao P, Pennells L, Kaptoge S, Caslake M, et al. Lipidrelated markers and cardiovascular disease prediction. Jama. 2012; 307(23):2499–506. Epub 2012/07/ 17. https://doi.org/10.1001/jama.2012.6571 PMID: 22797450; PubMed Central PMCID: PMC4211641.
- Holmes MV, Asselbergs FW, Palmer TM, Drenos F, Lanktree MB, Nelson CP, et al. Mendelian randomization of blood lipids for coronary heart disease. Eur Heart J. 2015; 36(9):539–50. Epub 2014/01/30. https://doi.org/10.1093/eurhearti/eht571 PMID: 24474739; PubMed Central PMCID: PMC4344957.
- Silverman MG, Ference BA, Im K, Wiviott SD, Giugliano RP, Grundy SM, et al. Association Between Lowering LDL-C and Cardiovascular Risk Reduction Among Different Therapeutic Interventions: A Systematic Review and Meta-analysis. Jama. 2016; 316(12):1289–97. Epub 2016/09/28. <u>https://doi.org/</u> 10.1001/jama.2016.13985 PMID: 27673306.
- Helgadottir A, Gretarsdottir S, Thorleifsson G, Hjartarson E, Sigurdsson A, Magnusdottir A, et al. Variants with large effects on blood lipids and the role of cholesterol and triglycerides in coronary disease. Nature genetics. 2016; 48(6):634–9. Epub 2016/05/03. <u>https://doi.org/10.1038/ng.3561</u> PMID: 27135400.
- Khera AV, Won HH, Peloso GM, O'Dushlaine C, Liu D, Stitziel NO, et al. Association of Rare and Common Variation in the Lipoprotein Lipase Gene With Coronary Artery Disease. Jama. 2017; 317(9):937–46. Epub 2017/03/08. https://doi.org/10.1001/jama.2017.0972 PMID: 28267856; PubMed Central PMCID: PMC5664181.
- Nordestgaard BG, Benn M, Schnohr P, Tybjaerg-Hansen A. Nonfasting triglycerides and risk of myocardial infarction, ischemic heart disease, and death in men and women. Jama. 2007; 298(3):299–308. Epub 2007/07/20. https://doi.org/10.1001/jama.298.3.299 PMID: 17635890.
- 53. Varbo A, Benn M, Tybjaerg-Hansen A, Jorgensen AB, Frikke-Schmidt R, Nordestgaard BG. Remnant cholesterol as a causal risk factor for ischemic heart disease. J Am Coll Cardiol. 2013; 61(4):427–36. Epub 2012/12/26. https://doi.org/10.1016/j.jacc.2012.08.1026 PMID: 23265341.
- 54. Marston NA, Giugliano RP, Im K, Silverman MG, O'Donoghue ML, Wiviott SD, et al. Association Between Triglyceride Lowering and Reduction of Cardiovascular Risk Across Multiple Lipid-Lowering Therapeutic Classes: A Systematic Review and Meta-Regression Analysis of Randomized Controlled Trials. Circulation. 2019; 140(16):1308–17. Epub 2019/09/19. https://doi.org/10.1161/ CIRCULATIONAHA.119.041998 PMID: 31530008; PubMed Central PMCID: PMC6791781.
- 55. Voight BF, Peloso GM, Orho-Melander M, Frikke-Schmidt R, Barbalic M, Jensen MK, et al. Plasma HDL cholesterol and risk of myocardial infarction: a mendelian randomisation study. Lancet. 2012; 380 (9841):572–80. Epub 2012/05/23. https://doi.org/10.1016/S0140-6736(12)60312-2 PMID: 22607825; PubMed Central PMCID: PMC3419820.
- 56. Emerging Risk Factors C, Sarwar N, Gao P, Seshasai SR, Gobin R, Kaptoge S, et al. Diabetes mellitus, fasting blood glucose concentration, and risk of vascular disease: a collaborative meta-analysis of 102 prospective studies. Lancet. 2010; 375(9733):2215–22. Epub 2010/07/09. https://doi.org/10.1016/S0140-6736(10)60484-9 PMID: 20609967; PubMed Central PMCID: PMC2904878.
- Le Goff W, Guerin M, Chapman MJ. Pharmacological modulation of cholesteryl ester transfer protein, a new therapeutic target in atherogenic dyslipidemia. Pharmacol Ther. 2004; 101(1):17–38. Epub 2004/ 01/20. https://doi.org/10.1016/j.pharmthera.2003.10.001 PMID: 14729390.
- Schwartz GG, Olsson AG, Abt M, Ballantyne CM, Barter PJ, Brumm J, et al. Effects of dalcetrapib in patients with a recent acute coronary syndrome. N Engl J Med. 2012; 367(22):2089–99. Epub 2012/11/ 07. https://doi.org/10.1056/NEJMoa1206797 PMID: 23126252.
- Lincoff AM, Nicholls SJ, Riesmeyer JS, Barter PJ, Brewer HB, Fox KAA, et al. Evacetrapib and Cardiovascular Outcomes in High-Risk Vascular Disease. N Engl J Med. 2017; 376(20):1933–42. Epub 2017/ 05/18. https://doi.org/10.1056/NEJMoa1609581 PMID: 28514624.
- Teslovich TM, Musunuru K, Smith AV, Edmondson AC, Stylianou IM, Koseki M, et al. Biological, clinical and population relevance of 95 loci for blood lipids. Nature. 2010; 466(7307):707–13. Epub 2010/08/06. https://doi.org/10.1038/nature09270 PMID: 20686565; PubMed Central PMCID: PMC3039276.
- Thompson A, Di Angelantonio E, Sarwar N, Erqou S, Saleheen D, Dullaart RP, et al. Association of cholesteryl ester transfer protein genotypes with CETP mass and activity, lipid levels, and coronary risk. Jama. 2008; 299(23):2777–88. Epub 2008/06/19. <u>https://doi.org/10.1001/jama.299.23.2777</u> PMID: 18560005.

- Group HTRC, Bowman L, Hopewell JC, Chen F, Wallendszus K, Stevens W, et al. Effects of Anacetrapib in Patients with Atherosclerotic Vascular Disease. N Engl J Med. 2017; 377(13):1217–27. Epub 2017/08/30. https://doi.org/10.1056/NEJMoa1706444 PMID: 28847206.
- Ference BA, Kastelein JJP, Ginsberg HN, Chapman MJ, Nicholls SJ, Ray KK, et al. Association of Genetic Variants Related to CETP Inhibitors and Statins With Lipoprotein Levels and Cardiovascular Risk. Jama. 2017; 318(10):947–56. Epub 2017/08/29. https://doi.org/10.1001/jama.2017.11467 PMID: 28846118; PubMed Central PMCID: PMC5710502.
- 64. Finan C, Gaulton A, Kruger FA, Lumbers RT, Shah T, Engmann J, et al. The druggable genome and support for target identification and validation in drug development. Sci Transl Med. 2017; 9(383). Epub 2017/03/31. https://doi.org/10.1126/scitranslmed.aag1166 PMID: 28356508; PubMed Central PMCID: PMC6321762.
- Sirota M, Schaub MA, Batzoglou S, Robinson WH, Butte AJ. Autoimmune disease classification by inverse association with SNP alleles. PLoS Genet. 2009; 5(12):e1000792. Epub 2009/12/31. https://doi.org/10.1371/journal.pgen.1000792 PMID: 20041220; PubMed Central PMCID: PMC2791168.
- 66. Gleason KJ, Yang F, Pierce BL, He X, Chen LS. Primo: integration of multiple GWAS and omics QTL summary statistics for elucidation of molecular mechanisms of trait-associated SNPs and detection of pleiotropy in complex traits. Genome Biol. 2020; 21(1):236. Epub 2020/09/12. https://doi.org/10.1186/s13059-020-02125-w PMID: 32912334; PubMed Central PMCID: PMC7488447.
- Sulc JA, Sonrel A, Mounier N, Auwerx C, Marouli E, Darrous L, et al. Composite trait Mendelian Randomization reveals distinct metabolic and lifestyle consequences of differences in body shape. medRxiv. 2020:2020.09.03.20187567. https://doi.org/10.1101/2020.09.03.20187567
- Dahl A, Cai N, Ko A, Laakso M, Pajukanta P, Flint J, et al. Reverse GWAS: Using genetics to identify and model phenotypic subtypes. PLoS genetics. 2019; 15(4):e1008009. Epub 2019/04/06. https://doi. org/10.1371/journal.pgen.1008009 PMID: 30951530; PubMed Central PMCID: PMC6469799.
- 69. King EA, Davis JW, Degner JF. Are drug targets with genetic support twice as likely to be approved? Revised estimates of the impact of genetic support for drug mechanisms on the probability of drug approval. Plos Genetics. 2019; 15(12). WOS:000512336600013. https://doi.org/10.1371/journal.pgen. 1008489 PMID: 31830040
- Nelson MR, Tipney H, Painter JL, Shen J, Nicoletti P, Shen Y, et al. The support of human genetic evidence for approved drug indications. Nat Genet. 2015; 47(8):856–60. Epub 2015/06/30. <u>https://doi.org/10.1038/ng.3314</u> PMID: 26121088.
- Nordestgaard BG, Varbo A. Triglycerides and cardiovascular disease. Lancet. 2014; 384(9943):626– 35. Epub 2014/08/19. https://doi.org/10.1016/S0140-6736(14)61177-6 PMID: 25131982.
- 72. Madsen CM, Varbo A, Nordestgaard BG. Extreme high high-density lipoprotein cholesterol is paradoxically associated with high mortality in men and women: two prospective cohort studies. Eur Heart J. 2017; 38(32):2478–86. Epub 2017/04/19. https://doi.org/10.1093/eurheartj/ehx163 PMID: 28419274.
- 73. Ko DT, Alter DA, Guo H, Koh M, Lau G, Austin PC, et al. High-Density Lipoprotein Cholesterol and Cause-Specific Mortality in Individuals Without Previous Cardiovascular Conditions: The CANHEART Study. J Am Coll Cardiol. 2016; 68(19):2073–83. Epub 2016/11/05. https://doi.org/10.1016/j.jacc.2016. 08.038 PMID: 27810046.
- 74. Guerin M, Silvain J, Gall J, Darabi M, Berthet M, Frisdal E, et al. Association of Serum Cholesterol Efflux Capacity With Mortality in Patients With ST-Segment Elevation Myocardial Infarction. J Am Coll Cardiol. 2018; 72(25):3259–69. Epub 2018/12/24. https://doi.org/10.1016/j.jacc.2018.09.080 PMID: 30573028.
- 75. Schizophrenia Working Group of the Psychiatric Genomics C. Biological insights from 108 schizophrenia-associated genetic loci. Nature. 2014; 511(7510):421–7. Epub 2014/07/25. https://doi.org/10.1038/ nature13595 PMID: 25056061; PubMed Central PMCID: PMC4112379.
- 76. Pickrell J, Berisa T, Segurel L, Tung JY, Hinds D. Detection and interpretation of shared genetic influences on 40 human traits. bioRxiv. 2015.
- Makalic E, Schmidt DF. An efficient algorithm for sampling from sin k (x) for generating random correlation matrices. Communications in Statistics-Simulation and Computation. 2020:1–5.
- Genomes Project C, Auton A, Brooks LD, Durbin RM, Garrison EP, Kang HM, et al. A global reference for human genetic variation. Nature. 2015; 526(7571):68–74. Epub 2015/10/04. <u>https://doi.org/10.1038/nature15393</u> PMID: 26432245; PubMed Central PMCID: PMC4750478.
- 79. Berisa T, Pickrell JK. Approximately independent linkage disequilibrium blocks in human populations. Bioinformatics. 2016; 32(2):283–5. Epub 2015/09/24. https://doi.org/10.1093/bioinformatics/btv546 PMID: 26395773; PubMed Central PMCID: PMC4731402.
- Julienne H, Zoufir A, Audit B, Arneodo A. Human genome replication proceeds through four chromatin states. PLoS Comput Biol. 2013; 9(10):e1003233. Epub 2013/10/17. https://doi.org/10.1371/journal. pcbi.1003233 PMID: 24130466; PubMed Central PMCID: PMC3794905.

- Natarajan A, Yardimci GG, Sheffield NC, Crawford GE, Ohler U. Predicting cell-type-specific gene expression from regions of open chromatin. Genome Res. 2012; 22(9):1711–22. Epub 2012/09/08. https://doi.org/10.1101/gr.135129.111 PMID: 22955983; PubMed Central PMCID: PMC3431488.
- Backenroth D, He Z, Kiryluk K, Boeva V, Pethukova L, Khurana E, et al. FUN-LDA: A Latent Dirichlet Allocation Model for Predicting Tissue-Specific Functional Effects of Noncoding Variation: Methods and Applications. American Journal of Human Genetics. 2018; 102:920–42. <u>https://doi.org/10.1016/j.ajhg.</u> 2018.03.026 PMID: 29727691
- **83.** Ghahramani Z, Jordan MI, editors. Supervised learning from incomplete data via an EM approach. Advances in neural information processing systems; 1994.
- Purcell S, Neale B, Todd-Brown K, Thomas L, Ferreira MA, Bender D, et al. PLINK: a tool set for wholegenome association and population-based linkage analyses. American journal of human genetics. 2007; 81(3):559–75. Epub 2007/08/19. <u>https://doi.org/10.1086/519795</u> PMID: <u>17701901</u>; PubMed Central PMCID: PMC1950838.
- Rousseeuw P J. Silhouettes: A graphical aid to the interpretation and validation of cluster analysis. Journal of Computational and Applied Mathematics. 1987; 20:53–65.
- Watanabe K, Taskesen E, Van Bochoven A, Posthuma D. Functional mapping and annotation of genetic associations with FUMA. Nature Communications. 2017; 8:1–10. https://doi.org/10.1038/ s41467-016-0009-6 PMID: 28232747.
- Kanehisa M, Goto S. KEGG: kyoto encyclopedia of genes and genomes. Nucleic acids research. 2000; 28(1):27–30. Epub 1999/12/11. <u>https://doi.org/10.1093/nar/28.1.27</u> PMID: <u>10592173</u>; PubMed Central PMCID: PMC102409.
- The Gene Ontology C. The Gene Ontology Resource: 20 years and still GOing strong. Nucleic acids research. 2019; 47(D1):D330–D8. Epub 2018/11/06. <u>https://doi.org/10.1093/nar/gky1055</u> PMID: 30395331; PubMed Central PMCID: PMC6323945.
- Luo W, Brouwer C. Pathview: an R/Bioconductor package for pathway-based data integration and visualization. Bioinformatics. 2013; 29(14):1830–1. Epub 2013/06/07. https://doi.org/10.1093/bioinformatics/btt285 PMID: 23740750; PubMed Central PMCID: PMC3702256.